Immobilisation and functional treatment for acute lateral ankle ligament injuries in adults (Review)

Kerkhoffs GMMJ, Rowe BH, Assendelft WJJ, Kelly KD, Struijs PAA, van Dijk CN



This is a reprint of a Cochrane review, prepared and maintained by The Cochrane Collaboration and published in *The Cochrane Library* 2009, Issue 1

http://www.thecochranelibrary.com



TABLE OF CONTENTS

HEADER	1
ABSTRACT	1
PLAIN LANGUAGE SUMMARY	2
BACKGROUND	2
OBJECTIVES	3
МЕТНОДЯ	3
RESULTS	4
DISCUSSION	7
AUTHORS' CONCLUSIONS	8
ACKNOWLEDGEMENTS	8
REFERENCES	8
CHARACTERISTICS OF STUDIES	12
DATA AND ANALYSES	29
Analysis 1.1. Comparison 1 Immobilisation vs. functional treatment, Outcome 1 Numbers not returning to sports.	32
Analysis 1.2. Comparison 1 Immobilisation vs. functional treatment, Outcome 2 Numbers not returning to sports.	33
Analysis 1.2. Comparison 1 Immobilisation vs. functional treatment, Outcome 2 Pain.	34
Analysis 1.3. Comparison 1 Immobilisation vs. functional treatment, Outcome 5 Pain	34 35
Analysis 1.4. Comparison 1 Immobilisation vs. functional treatment, Outcome 4 Sweiling.	
	36
Analysis 1.6. Comparison 1 Immobilisation vs. functional treatment, Outcome 6 Objective instability (talar tilt and/or	
ADS)	37
Analysis 1.7. Comparison 1 Immobilisation vs. functional treatment, Outcome 7 Recurrent sprain.	38
Analysis 1.8. Comparison 1 Immobilisation vs. functional treatment, Outcome 8 Decreased ROM.	39
Analysis 1.9. Comparison 1 Immobilisation vs. functional treatment, Outcome 9 Patient satisfaction	40
Analysis 1.10. Comparison 1 Immobilisation vs. functional treatment, Outcome 10 Return to sports (days).	41
Analysis 1.11. Comparison 1 Immobilisation vs. functional treatment, Outcome 11 Return to work (days)	42
Analysis 1.12. Comparison 1 Immobilisation vs. functional treatment, Outcome 12 Improvement in pain score.	42
Analysis 1.13. Comparison 1 Immobilisation vs. functional treatment, Outcome 13 Improvement in swelling	43
Analysis 1.14. Comparison 1 Immobilisation vs. functional treatment, Outcome 14 Improvement in objective instability	62
(difference in TT).	43
Analysis 1.15. Comparison 1 Immobilisation vs. functional treatment, Outcome 15 Improvement in ROM.	44
Analysis 2.1. Comparison 2 Immobilisation vs. physiotherapy, Outcome 1 Return to work (days).	44
Analysis 3.1. Comparison 3 Immobilisation vs. other types of immobilisation, Outcome 1 Pain.	45
Analysis 3.2. Comparison 3 Immobilisation vs. other types of immobilisation, Outcome 2 Swelling.	45
Analysis 3.3. Comparison 3 Immobilisation vs. other types of immobilisation, Outcome 3 Objective instability (talar tilt	
and/or ADS)	46
Analysis 3.4. Comparison 3 Immobilisation vs. other types of immobilisation, Outcome 4 Return to work (days).	46
Analysis 4.1. Comparison 4 Immobilisation vs. functional treatment (high quality trials), Outcome 1 Numbers not	
returning to sports.	47
Analysis 4.2. Comparison 4 Immobilisation vs. functional treatment (high quality trials), Outcome 2 Numbers not	
returning to work.	48
Analysis 4.3. Comparison 4 Immobilisation vs. functional treatment (high quality trials), Outcome 3 Pain.	48
Analysis 4.4. Comparison 4 Immobilisation vs. functional treatment (high quality trials), Outcome 4 Swelling.	50
Analysis 4.5. Comparison 4 Immobilisation vs. functional treatment (high quality trials), Outcome 5 Subjective instability	
(giving way).	51
Analysis 4.6. Comparison 4 Immobilisation vs. functional treatment (high quality trials), Outcome 6 Objective instability	71
(talar tilt and/or ADS).	52
Analysis 4.7. Comparison 4 Immobilisation vs. functional treatment (high quality trials), Outcome 7 Recurrent sprain.	52 53
	53 54
Analysis 4.8. Comparison 4 Immobilisation vs. functional treatment (high quality trials), Outcome 8 Patient satisfaction.	54
Analysis 4.9. Comparison 4 Immobilisation vs. functional treatment (high quality trials), Outcome 9 Return to work	- /
(days)	54
Immobilisation and functional treatment for acute lateral ankle ligament injuries in adults (Review)	i

Analysis 4.10. Comparison 4 Immobilisation vs. functional treatment (high quality trials), Outcome 10 Improvement in	
pain score	55
Analysis 4.11. Comparison 4 Immobilisation vs. functional treatment (high quality trials), Outcome 11 Improvement in	
∂	55
Analysis 4.12. Comparison 4 Immobilisation vs. functional treatment (high quality trials), Outcome 12 Improvement in	
ROM	56
ADDITIONAL TABLES	56
APPENDICES	57
WHAT'S NEW	58
HISTORY	58
CONTRIBUTIONS OF AUTHORS	58
DECLARATIONS OF INTEREST	58
SOURCES OF SUPPORT	59
INDEX TERMS	59

[Intervention Review]

Immobilisation and functional treatment for acute lateral ankle ligament injuries in adults

Gino MMJ Kerkhoffs¹, Brian H Rowe², Willem JJ Assendelft³, Karen D Kelly⁴, Peter AA Struijs¹, C N van Dijk⁵

¹Department of Orthopaedic Surgery, Academic Medical Center, Amsterdam, Netherlands. ²Department of Emergency Medicine, University of Alberta, Edmonton, Canada. ³Department of Public Health and Primary Care, Leiden University Medical Centre, Leiden, Netherlands. ⁴Faculty of Nursing, University of Alberta, Edmonton, Canada. ⁵Department of Orthopaedic Surgery, Academic Medical Center, Amsterdam, Netherlands

Contact address: Gino MMJ Kerkhoffs, Department of Orthopaedic Surgery, Academic Medical Center, University of Amsterdam, Meibergdreef 9, Amsterdam, 1105 AZ, Netherlands. g.m.kerkhoffs@amc.uva.nl. gino.kerkhoffs@kssg.ch.

Editorial group: Cochrane Bone, Joint and Muscle Trauma Group. Publication status and date: Edited (no change to conclusions), published in Issue 1, 2009. Review content assessed as up-to-date: 13 May 2002.

Citation: Kerkhoffs GMMJ, Rowe BH, Assendelft WJJ, Kelly KD, Struijs PAA, van Dijk CN. Immobilisation and functional treatment for acute lateral ankle ligament injuries in adults. *Cochrane Database of Systematic Reviews* 2002, Issue 3. Art. No.: CD003762. DOI: 10.1002/14651858.CD003762.

Copyright © 2009 The Cochrane Collaboration. Published by John Wiley & Sons, Ltd.

ABSTRACT

Background

Acute lateral ankle ligament injuries (ankle sprains) are common problems in acute medical care. The treatment variation observed for the acutely injured lateral ankle ligament complex suggests a lack of evidence-based management strategies for this problem.

Objectives

The objective of this review was to assess the effectiveness of methods of immobilisation for acute lateral ankle ligament injuries and to compare immobilisation with functional treatment methods.

Search methods

We searched the Cochrane Bone, Joint and Muscle Trauma Group specialised register (December 2001); the Cochrane Controlled Trials Register (*The Cochrane Library*, Issue 4, 2001), MEDLINE (1966-May 2000), EMBASE (1988-May 2000), reference lists of articles, and contacted organisations and researchers in the field.

Selection criteria

Randomised and quasi-randomised controlled trials comparing either different types of immobilisation or immobilisation versus functional treatments for injuries to the lateral ligament complex of the ankle in adults were included. Trials which investigated the treatment of chronic instability or post-surgical treatment were excluded.

Data collection and analysis

Data were independently extracted by two authors. Where appropriate, results of comparable studies were pooled using fixed effects models. Individual and pooled statistics were reported as relative risks with 95% confidence intervals for dichotomous outcomes and weighted (WMD) or standardised (SMD) mean differences and 95% confidence intervals for continuous outcome measures. Heterogeneity between trials was tested using a standard chi-squared test.

Main results

Twenty-one trials involving 2184 participants were included. The mean validity score of the included trials increased from 9.1 (SD 3.0) to 10 (SD 2.9) after retrieving further information (maximum 18 points). Statistically significant differences in favour of functional treatment when compared with immobilisation were found for seven outcome measures: more patients returned to sport in the long term (relative risk (RR) 1.86, 95% confidence interval (CI) 1.22 to 2.86); the time taken to return to sport was shorter (WMD 4.88 (days), 95% CI 1.50 to 8.25); more patients had returned to work at short term follow-up (RR 5.75, 95% CI 1.01 to 32.71); the time taken to return to work was shorter (WMD 8.23 days, 95% CI 6.31 to 10.16); fewer patients suffered from persistent swelling at short term follow-up (RR 1.74, 95% CI 1.17 to 2.59); fewer patients suffered from objective instability as tested by stress X-ray (WMD 2.60, 95% CI 1.24 to 3.96); and patients treated functionally were more satisfied with their treatment (RR 1.83, 95% CI 1.09 to 3.07). A separate analysis of trials that scored 50 per cent or more in quality assessment found a similar result for time to return to work only (WMD (days) 12.89, 95% CI 7.10 to 18.67). No significant differences between varying types of immobilisation, immobilisation and physiotherapy or no treatment were found, apart from one trial where patients returned to work sooner after treatment with a soft cast. In all analyses performed, no results were significantly in favour of immobilisation.

Authors' conclusions

Functional treatment appears to be the favourable strategy for treating acute ankle sprains when compared with immobilisation. However, these results should be interpreted with caution, as most of the differences are not significant after exclusion of the low quality trials. Many trials were poorly reported and there was variety amongst the functional treatments evaluated.

PLAIN LANGUAGE SUMMARY

Immobilisation and functional treatment for acute lateral ankle ligament injuries in adults

Ankle sprains are one of the most common injuries of active people. They are usually treated by either a plaster cast being placed around the ankle so that the joint cannot move, or by treatments that only support the ankle. These are known as functional treatments and can include tapes, bandages or wraps. This review of trials found that functional treatment helped patients return to work and sports more quickly, and helped reduce swelling initially. People were more satisfied with functional treatment. There were no differences between treatments for pain, how easily the ankle could move after treatment, or whether it was likely that another sprain would happen.

BACKGROUND

Injuries to the lateral ligament complex of the ankle are common problems in acute care practice. It is estimated that one ankle sprain occurs per 10,000 population per day (Katcherian 1994). Overall, injuries of the lateral ligament complex of the ankle form a quarter of all sports injuries (Keeman 1990). Some sports (e.g. basketball, soccer and volleyball) have a particularly high incidence of ankle injuries (Lindenfeld 1994; Luidinga 1985). The treatment of inversion injuries is performed by emergency and primary health care physicians as well as by orthopaedic and trauma surgeons (Kannus 1991). The total annual costs to society for ankle injuries has been estimated to be approximately 40 million Euro per one million people (Zeegers 1995).

The nomenclature for lesions of the lateral ligament complex of the ankle is variable. Many terms are applied to the injured ligament such as ankle sprain or ankle distortion. Most authors use the term 'sprain' to describe a morphologic condition, representing a diversity of pathology, ranging from overstretching of the ligament to complete rupture with instability of the joint (Watson-Jones 1976). To classify the severity of the lateral ankle ligament injuries a grading system from I to III has been introduced (Bernett 1979; Marti 1982; Kannus 1991; Lassiter 1989; van Dijk 1994). Grade I is a mild stretching of the ligament with no instability, grade II is a partial rupture with mild instability of the joint (such as isolated rupture of the anterior talofibular ligament) and grade III involves complete rupture of the ligaments with instability of the joint.

The most common mechanism of injury is supination and adduction (usually referred to as inversion) of the plantar-flexed foot. It is known that the anterior talofibular ligament is the first or only ligament to sustain injury in 97 per cent of cases (Brostrom 1965; van Dijk 1994). Brostrom (Brostrom 1965) found that combined

ruptures of the anterior talofibular ligament and the calcaneofibular ligament occurred in 20 per cent of the cases and that isolated rupture of the calcaneofibular ligament occurs in only three per cent. The posterior talofibular ligament is usually uninjured unless there is a frank dislocation of the ankle. Together, these three ligaments (anterior talofibular, calcaneofibular, posterior talofibular) form the lateral ligament complex of the ankle (Wiersma 1998).

Many different treatments are used for acute ankle sprain. The three main modalities of treatment are: 1) operative treatment; 2) conservative treatment with plaster cast immobilisation and 3) functional treatment. The latter is an early mobilisation programme and involves the use of an external support (e.g. tape and/ or elastic bandage or orthotic support), combined with co-ordination training.

The treatment practice variation identified for lateral ankle ligament complex injuries suggests a lack of evidence-based management strategies for this problem (Brostrom 1966; Brostrom 1966a; van Dijk 1994; van Moppes 1982). Dehne (Dehne 1933) first reported ankle injury treatment with immobilisation below the knee. Many studies presenting results of this type of immobilisation have since been published (Alder 1976; Leonard 1949). Freeman (Freeman 1965b; Freeman 1965c) introduced a new concept in the conservative treatment of ruptures of the lateral ligaments of the ankle by suggesting that the use of proprioceptive training using co-ordination exercises could reduce the proprioceptive deficit and symptoms of the ankle "giving way". Consequently, many patients were treated functionally with supportive elastic bandage combined with co-ordination training. Functional treatment with tape bandage or orthotic support has become more popular in the last two decades (Jacob 1986; Leonard 1949; Moller-Larsen 1988; Vaes 1985; Stover 1980). The use of laser therapy, ultrasound treatment and/or acupuncture have all been reported, but none are used widely.

Despite all of these options, it is unclear which treatment is most appropriate. Those in favour of functional treatment cite advantages such as lower cost and decreased morbidity with the same probability of ankle stability when compared to operative treatment. However, underestimating the injury severity may lead to chronic instability of the lateral ankle ligament complex. Therefore, the treatment approach to these problems is important to clarify.

Using evidence from randomised controlled trials, this review evaluated the effectiveness of the various methods of immobilisation for acute ankle sprain against each other and functional treatment.

OBJECTIVES

The objectives of this review were to assess the effectiveness of methods of immobilisation for acute lateral ligament injury and to compare immobilisation with functional treatment strategies. The specific null hypotheses were:

1. No difference exists in outcome between different types or durations of immobilisation for treatment of acute injuries of the lateral ankle ligament complex.

2. No difference exists in outcome between any method of immobilisation and physiotherapy for treatment of acute injuries of the lateral ankle ligament complex.

3. No difference exists in outcome between any method of immobilisation and no intervention for treatment of acute injuries of the lateral ankle ligament complex.

4. No difference exists in outcome between any method of immobilisation and any method of functional treatment for acute injuries of the lateral ankle ligament complex.

The comparison of different types of functional treatment and the comparison of immobilisation with operative treatment for acute injuries of the lateral ankle ligament complex have been undertaken in separate reviews (Kerkhoffs 2002a; Kerkhoffs 2002b).

METHODS

Criteria for considering studies for this review

Types of studies

All randomised and quasi-randomised (methods of allocating participants to a treatment which are not strictly random e.g. date of birth, hospital record number or alternation) controlled trials comparing immobilisation with either another type or duration of immobilisation, or a functional treatment, for injuries to the lateral ligament complex of the ankle.

Types of participants

Studies enrolling skeletally mature individuals who reported an acute injury to the lateral ligament complex of the ankle were eligible for inclusion. The diagnosis could be based on either physical examination (positive anterior drawer test, pain and haematoma), a stress radiograph or an arthrogram of the injured ankle. Trials dealing exclusively with children (where growth plate injuries predominate), patients with congenital deformities or patients with degenerative conditions were excluded. A priori, we decided a mixed population of adults and children could be included if the adult population could be analysed separately, or the proportion of children was small (< 10%).

Trials which focussed on the treatment of chronic instability or post-surgical treatment were excluded. Patients with chronic instability have symptoms of pain, swelling, recurrent sprains and

instability for longer than six months (Karlsson 1997). If studies included participants with chronic ankle sprains or other ankle injuries such as avulsion fractures, then results from these studies were included in the review provided such injuries occurred in under ten per cent of the whole study population.

Types of interventions

Intervention:

Immobilisation, either by plaster cast or special boots.

Comparison:

a) physiotherapy;

b) functional interventions (including: elastic bandage, softcast, tape or orthosis with associated co-ordination training);c) non-intervention.

Types of outcome measures

The following outcomes were sought and extracted:

- 1. Return to pre-injury level of sports (yes/no; time to achieve)
- 2. Return to pre-injury level of work (yes/no; time to achieve)
- 3. Pain (yes/no) (continuous data)
- 4. Swelling (yes/no)
- 5. Subjective instability (e.g. 'giving way') (yes/no)

6. Objective instability (e.g. anterior drawer measures, talar tilt) (yes/no)

- 7. Recurrent injury (yes/no)
- 8. Ankle mobility/range of motion (continuous data)

Complications (e.g. sensory deficit, infection, arthrosis, osteoarthritis, allergic reaction, stiffness, muscle atrophy) (yes/no)
Patient satisfaction (ordinal, continuous or dichotomous data)
Follow-up times were grouped into:

A. Short term - within six weeks of randomisation (to identify early significant complications);

B. Intermediate term - six weeks to one year follow-up;

C. Long term - one to two years after treatment.

Search methods for identification of studies

We searched the Cochrane Bone, Joint and Muscle Trauma Group specialised register (December 2001), the Cochrane Controlled Trials Register (*The Cochrane Library*, Issue 4, 2001), MEDLINE (1966-May 2000), EMBASE (1988-May 2000) and reference lists of articles. Papers outside the English language were considered if translation was possible. We also contacted researchers in the field and the Medical Departments of the Dutch Defence Forces and the Royal Dutch Football Association.

In MEDLINE (OVID Web), the subject specific search was combined with the first two levels of the optimum search strategy (Clarke 2001) (see Appendix 1).

Data collection and analysis

Selection of studies

From the title, abstract, or descriptors, three reviewers (GK, PS, CVD) independently reviewed literature searches to identify potentially relevant trials for full review. From the full text, using the above criteria, two reviewers (BR, KK) independently selected trials for inclusion in this review (Dickersin 1992). Disagreement was resolved by consensus or third party adjudication (GK).

Data extraction and management

Data were independently extracted by two review authors (GK, PS) using a pre-piloted data extraction tool. After consensus, there was no disagreement and therefore no third party adjudication was necessary.

Where appropriate, results of similar studies were pooled using fixed effects models, after consideration of the heterogeneity between the trials. Individual and pooled statistics were reported as relative risks (RR) with 95 per cent confidence intervals (CI) for dichotomous outcomes and weighted or standardised mean differences and 95 per cent CI for continuous outcomes. Heterogeneity between trials was tested using a standard chi-squared test.

Assessment of risk of bias in included studies

In this review, risk of bias is implicitly assessed in terms of methodological quality.

Methodological quality for each study was independently assessed, without masking (Jadad 1996; Schulz 1994; Verhagen 1998), by two reviewers (BR, KK) from the group using a piloted, subjectspecific modification of the generic evaluation tool used by the Cochrane Bone, Joint and Muscle Trauma Group. Any disagreement was resolved by consensus or third party adjudication (GK). The scoring scheme for the 11 aspects of internal and external validity covered by this tool is given in Table 1. Our cut-off point for high and low-quality trials was arbitrarily set at 50 per cent of the maximum score.

RESULTS

Description of studies

See: Characteristics of included studies; Characteristics of excluded studies.

Fifty potentially eligible trials were identified from the electronic database search and their full texts retrieved. Independent review of these texts resulted in the inclusion of 21 trials and exclusion of

22 trials. The main reason for exclusion was that the intervention of interest was not described. For a more detailed description of the excluded studies, see the Characteristics of Excluded Studies Table.

Seven trials (Rocinski 1991; Grasmueck 1997; Hoogenband 1984; Duwairi 1998; Soosai Nathan 1997; Vitellas 1995; Zwipp 1986) have been placed in Studies Awaiting Assessment. These trials remain potentially eligible for inclusion in our review.

The 21 included studies enrolled a total of 2184 participants. Four trials investigated more than one comparison. Most included trials compared a rigid cast (with or without supplementary treatment at a later time such as a walking cast, braces or physio-therapy) with a form of functional treatment. The latter included semi-rigid casts (Avci 1998); tubigrips (Brakenbury 1983; Brooks 1981); bandages (Lind 1984); elastic bandages (Brostrom 1966; Korkala 1987); taping (Caro 1964; Moller-Larsen 1988; Sommer 1993); bracing (Cetti 1984; Dettori 1994; Klein 1991; Konradsen 1991; Milford 1990; Sommer 1993); wrapping (Gronmark 1978; Hedges 1980; Munk 1995; Roycroft 1983); strapping and mobilisation (Freeman 1965a); immobilisation without a cast (Brooks 1981); and physiotherapy (Brooks 1981).

One study (Eiff 1994) investigated splintage and crutches versus an elastic wrap and another (Regis 1995) compared an immobilisation gutter and weightbearing cast with a group which also used a dynamic brace for a period of time.

In general, participants were likely to be young (<50 years), and trials tended to include a higher percentage of males. Participant numbers ranged from 400 (Brakenbury 1983) to 20 (Regis 1995), with most enrolling 120 or less. For further details of individual included studies, see the Characteristics of Included Studies Table.

Risk of bias in included studies

The results of the methodological quality assessment are described in detail under 'Methods' in the Characteristics of Included Studies Table. The validity assessment comprised 11 items, each for a maximum of two points. However, it was not possible to ascribe points to items E and F in any of the trials, since blinding of patients and care providers proved to be impossible for the treatment strategies assessed. Therefore, a total of 18 points proved to be the maximum score attainable.

The initial agreement of the two reviewers on the quality assessment of the included trials was 86 per cent (163 out of 189 items). The Kappa value (K) for measurement of agreement beyond chance between these two reviewers was 0.78.

After the initial assessment, the quality score of included trials ranged from five to 18 points, with a mean score of 9.1 points (SD 3.0). Trials with a score 50 per cent and more of the maximum (11 and upwards) were categorized as 'high quality'. The initial assessment resulted in seven trials being scored as high quality (Eiff 1994; Klein 1991; Gronmark 1978; Munk 1995; Brostrom 1966; Caro 1964; Brooks 1981). After retrieving additional information

from the authors, the mean validity score increased to 10.4 (SD 2.9) and four trials (Avci 1998; Hedges 1980; Korkala 1987; Regis 1995) moved from being categorised as low quality to high quality. An additional six trials improved their score. A total of 11 studies (52%) involving 1098 patients were classified as high quality. After evaluating all results, an additional subgroup analysis was performed on the basis of trial quality.

Effects of interventions

Data were extracted on all relevant outcome measures as described above. Four types of comparisons are presented below;

A) Comparisons between different forms of immobilisation;

B) Comparisons between immobilisation and physiotherapy;

C) Comparisons between immobilisation and no treatment;

D) Comparisons between immobilisation and functional treatment.

A) DIFFERENT FORMS OF IMMOBILISATION

Two studies compared two different types of immobilisation of the ankle joint as a treatment of an acute ankle injury (Brooks 1981; Avci 1998). Avci 1998 compared a cast immobilisation with a semi-rigid cast immobilisation and Brooks 1981 compared cast immobilisation with immobilisation without a cast. At short term follow-up, both studies evaluated time to return to work. Avci 1998 reported a significantly shorter time to work return in the group treated with the semi-rigid cast (WMD 3.80 days, 95% CI 1.16 to 6.44). The results from Brooks 1981 could not be calculated.

Avci 1998 reported no significant differences at short term followup for pain, swelling or objective instability between the two immobilisation treatments.

B) IMMOBILISATION VERSUS PHYSIOTHERAPY

Only one study compared the results of immobilisation with physiotherapy (Brooks 1981). Two outcome measures, patient satisfaction and return to work, were reported. The percentage of patients returning to work at short term follow-up could not be calculated, because standard deviations were not reported nor could they be retrieved from the authors. Clear data was also not available for patient satisfaction, with Brooks 1981 reporting that "All patients that received physiotherapy were impressed with their treatment" and "All patients who had had their ankle immobilised in a plaster-of-Paris thought they had good pain relief."

C) IMMOBILISATION VERSUS NO TREATMENT

Only one study compared the results of immobilisation with no treatment (Brooks 1981). Patient satisfaction and return to work were reported. The percentage of patients returning to work at short term follow-up could not be calculated properly, because standard deviations were not reported nor could they be retrieved from the authors. Patient satisfaction was not described adequately and author contact failed to provide additional data.

D) IMMOBILISATION VERSUS FUNCTIONAL TREAT-MENT

Twenty trials described a type of immobilisation compared with a functional treatment. Seven trials compared immobilisation with use of an orthotic device, known as a brace (Cetti 1984; Dettori 1994; Klein 1991; Konradsen 1991; Milford 1990; Regis 1995; Sommer 1993); five trials compared immobilisation with elastic bandage or grip (Brakenbury 1983; Brooks 1981; Brostrom 1966; Korkala 1987; Lind 1984); four trials compared immobilisation with taping (Caro 1964; Freeman 1965a; Moller-Larsen 1988; Sommer 1993); and five trials compared cast immobilisation to treatment using a softcast or wrap (Eiff 1994; Gronmark 1978; Hedges 1980; Munk 1995; Roycroft 1983). Data were available for the following outcome measures:

1. Return to sports

A total of eight studies described return to sports as an outcome measure. The number of patients who had returned to sports activity at long term follow-up, pooled from five trials, (Korkala 1987; Regis 1995; Moller-Larsen 1988; Klein 1991; Eiff 1994), was significantly higher in the functional treatment compared with the immobilisation group (RR 1.86, 95%CI 1.22 to 2.86). Pooled results from three trials (Milford 1990; Freeman 1965a; Sommer 1993) demonstrate a significantly shorter time to return to sports for the functional treatment group (WMD 4.88 days, 95%CI 1.50 to 8.25).

2. Return to work

A total of nine studies reported on return to work as an outcome. The pooled results from six studies, (Brostrom 1966; Caro 1964; Brakenbury 1983; Cetti 1984; Dettori 1994; Roycroft 1983) demonstrated a significantly shorter time to return to work for the functionally treated group (WMD 8.23 days, 95% CI 6.31 to 10.16). The results of two studies (Eiff 1994; Konradsen 1991) showed a significantly higher percentage of patients had returned to work at short term follow-up in the functionally treated group compared with the immobilisation group (RR 5.75, 95%CI 1.01 to 32.71).

3. Pain

Nine studies reported whether the patient was experiencing pain after treatment. Three studies described short term results (Cetti 1984; Dettori 1994; Eiff 1994), five intermediate results (Cetti 1984; Dettori 1994; Eiff 1994; Hedges 1980; Lind 1984) and five long term results (Eiff 1994; Freeman 1965a; Gronmark 1978; Klein 1991; Munk 1995). No significant differences were found for any follow-up time. Only one study (Hedges 1980) evaluated pain as a continuous score and found no significant differences between the two treatments.

4. Swelling

Six trials reported swelling as an outcome measure (Brakenbury 1983; Freeman 1965a; Hedges 1980; Klein 1991; Eiff 1994; Cetti 1984). At short term follow-up, pooled results from Brakenbury 1983, Cetti 1984 and Eiff 1994 found significantly fewer patients suffered from persistent swelling of the ankle in the functionally treated group (RR 1.74, 95% CI 1.17 to 2.59). At intermediate and long term follow-up these differences were no longer observed.

As details of the validity of methods used to measure swelling were unavailable, this difference should be interpreted with caution. 5. Subjective instability or 'giving way'

'Giving way' is a subjective outcome measure to describe the amount of instability in the ankle joint. Seven studies reported this outcome measure (Brostrom 1966; Cetti 1984; Dettori 1994; Eiff 1994; Freeman 1965a; Klein 1991; Korkala 1987). No statistically significant differences were found at short, intermediate or long term follow-up.

6. Objective instability

Objective instability of the ankle joint is either measured using the Talar Tilt Test (TTT) or the Anterior Drawer Test (ADT). Seven trials described the outcomes of either the TTT or the ADT in order to evaluate the success of the two treatment modalities (Brooks 1981; Freeman 1965a; Konradsen 1991; Korkala 1987; Lind 1984; Munk 1995; Sommer 1993). Shortly after an inversion injury with ligament damage, TTT and ADT are almost always positive, hence results at short term are not relevant. Intermediate follow-up results (Sommer 1993) demonstrated a significantly increased instability using the TTT in patients treated with immobilisation, when described as a continuous outcome measure (WMD 2.60 degrees, 95% CI 1.24 to 3.96). Dichotomous results at intermediate (RR 0.69, 95% CI 0.14 to 3.34) as well as long term follow-up (RR 0.91, 95% CI 0.60 to 1.39) failed to demonstrate a significant difference.

7. Recurrent sprain

The frequency of a recurrent sprain is best measured at intermediate or long term follow-up. Eleven trials reported this outcome (Sommer 1993; Munk 1995; Regis 1995; Lind 1984; Klein 1991; Korkala 1987; Brostrom 1966; Cetti 1984; Dettori 1994; Eiff 1994; Konradsen 1991). No significant differences were demonstrated at any follow-up time.

8. Range of motion

An impaired range of motion (ROM) after inversion injury is likely the result of soft tissue swelling and pain. Three studies evaluated ROM as an outcome measure (Brakenbury 1983; Cetti 1984; Munk 1995). In most cases at intermediate or long term follow-up, ROM will be restored in almost all patients, therefore the analysis of ROM as an outcome in this case is mainly relevant at short term follow-up. Two studies (Brakenbury 1983, Cetti 1984) found no difference between treatment group (RR 1.27, 95% CI 0.73 to 2.19). At intermediate follow-up, Cetti 1984 found comparable results (RR 7.00, 95% CI 0.37 to 131.28).

9. Patient satisfaction

Six trials reported patient satisfaction (reported as a dichotomous outcome, satisfied yes/no) to evaluate results (Brooks 1981; Cetti 1984; Klein 1991; Moller-Larsen 1988; Munk 1995; Roycroft 1983). At intermediate follow-up, significantly more patients were satisfied with functional treatment (RR 4.25, 95% CI 1.12 to 16.09). After pooling the results of all studies, ignoring the follow-up period, more patients were satisfied with treatment in the functionally treated group compared with the immobilisation group

(RR 1.83, 95% CI 1.09 to 3.07).

In addition to the evaluation of these results, a sub-group analysis was performed including only those trials which obtained over 50 per cent of the possible methodological quality score.

A) IMMOBILISATION VERSUS FUNCTIONAL TREAT-MENT: HIGH QUALITY TRIALS

The 11 'high quality' trials (score 50% or more of the quality score) were analysed separately in order to compare the results with the overall results and subsequently determine if publication bias was introduced by low quality trials. The focus was restricted to the comparison of immobilisation versus functional treatment, since immobilisation versus physiotherapy or immobilisation versus an alternative type of immobilisation was already analysed and described by high quality trials only.

Return to sports

Four trials, (Eiff 1994; Klein 1991; Korkala 1987; Regis 1995), contributed to this analysis and a pooled analysis of long term results failed to show a significant difference between groups (RR 1.70, 95% CI 0.98 to 2.95).

Return to work

The time to return to work, as described by two trials (Brostrom 1966; Caro 1964) was shorter in the functionally treated group when results were pooled (WMD 12.89 days, 95% CI 7.10 to 18.67). For the numbers of patients returning to work at short, intermediate and long term follow-up, no significant differences were found.

Pain

Five trials categorised as high quality reported pain as an outcome measure (Eiff 1994; Gronmark 1978; Hedges 1980; Klein 1991; Munk 1995). No significant differences were seen at either short (RR 1.31, 95% CI 0.87 to 1.97), intermediate (RR 1.17, 95% CI 0.66 to 2.09) or long term follow-up (RR 1.23, 95% CI 0.71 to 2.14).

Swelling

Two high quality trials (Eiff 1994; Klein 1991) analysed persistent swelling as a dichotomous outcome measure. No statistically significant differences between the immobilisation group and the functionally treated group were identified at short, intermediate or long term follow-up.

ROM

One high quality trial (Munk 1995) described improvement in ROM as a continuous outcome measure, and found no statistically significant differences between the groups at long term follow-up (WMD -1.00, 95% CI -4.59 to 2.59).

Subjective instability

For short and intermediate term follow-up, only one high quality trial (Eiff 1994) reported this outcome. There were no significant differences between the treatments. Long term results from four trials (Brostrom 1966; Eiff 1994; Klein 1991; Korkala 1987) also failed to demonstrate any significant differences. Objective instability

Long term results on objective instability (Brostrom 1966; Korkala

1987; Munk 1995) showed no statistically significant differences. Recurrent sprains

Over the long term, there were no differences in the likelihood of a recurrent sprain after pooling the results of six high quality trials (Eiff 1994; Brostrom 1966; Klein 1991; Korkala 1987; Munk 1995; Regis 1995, RR 1.25, 95% CI 0.84 to 1.85). Patient satisfaction

Three high quality trials evaluated satisfaction as an outcome measure (Brooks 1981; Klein 1991; Munk 1995). After pooling the results of Klein 1991 and Munk 1995, no statistically significant differences were identified (RR 1.68, 95%CI 0.72 to 3.90).

DISCUSSION

The main objective of this review was to evaluate the effectiveness of the various types of immobilisation as a treatment for acute ankle sprain in adults, using evidence from randomised controlled trials. Twenty-one trials described a type of immobilisation in comparison to a functional treatment. Overall, there appeared to be better results with the functional treatment. No outcomes were shown to be improved by immobilisation treatment.

After evaluation of the results, statistically significant differences were found for seven outcome measures, all in favour of functional treatment: a higher percentage of patients returning to sports; the time to return to work was shorter; fewer patients suffered from persistent swelling; fewer patients suffered from objective instability at intermediate follow-up and range of motion was less limited in those patients treated functionally compared to patients treated with immobilisation. Overall, patients treated functionally were also more satisfied.

The quality of the included studies was acceptable. Our cut-off point for high and low-quality trials was arbitrarily set at 50 per cent of the maximum score. Given this cut-off point, a total of 11 studies (52%) were categorised for the purposes of this review as 'high quality'. A separate analysis of the high quality trials was performed; ten trials increased in quality assessment after we received additional information, which suggests poor reporting to be an important factor in initial validity assessment. When examining the effect of exclusion of lower rated trials, only the pooled result for time to return to work remained significant. However, in both analyses, no results were significantly in favour of immobilisation.

The above described results are in accordance with the current opinion that functional treatment is the treatment of choice for acute lateral ankle ligament injuries. Kannus and Renstrom (Kannus 1991b), Tiling (Tiling 1994), Ogilvie-Harris (Ogilvie-Harris 1995) and Shrier (Shrier 1995) all found functional treatment to be the most effective.

The primary objective of this review was met by our finding that immobilisation should no longer be the non-operative treatment of choice for patients suffering from an acute ankle sprain.

AUTHORS' CONCLUSIONS

Implications for practice

Functional treatment with early mobilisation appears to provide improved outcomes for patients compared with immobilisation.

The relative benefit of functional treatment versus surgical treatment, especially for severe ankle sprains, is beyond the scope of this review. Two further Cochrane reviews (Kerkhoffs 2002a; Kerkhoffs 2002b) present results for comparisons among different functional treatments, and for surgical interventions for acute ankle sprain.

Implications for research

Functional treatment currently seems an appropriate treatment. However, a great variety of functional treatments are described in trials included in this review. Readers are directed therefore to the Cochrane review (Kerkhoffs 2002a) which compares these strategies for this injury.

To enhance existing evidence for effectiveness of physiotherapy as a treatment strategy for acute ankle sprains, a well conducted randomised clinical trial with accurate description of methodology and adequate power is warranted.

ACKNOWLEDGEMENTS

We thank Helen Handoll and Leeann Morton for their help and advice. We would also like to thank the following for helpful comments at editorial review: William Gillespie, Peter Gotzsche, Gordon Murray, Rajan Madhok, Rene Marti, Marc Swiontkowski, Rocco Pitto, and Harley Gray. We thank the following authors for responding to our request for additional information: Peter Gotzsche, Lennart Brostrom, Jerris Hedges, Patrice Eiff, S. Brooks, Flemming Moller-Larsen, O. Korkola, Bo Munk, Michael Freeman, Dario Regis, Sinan Avci and S. Soosai Nathan.

REFERENCES

References to studies included in this review

Avci 1998 {published and unpublished data}

Avci S, Sayli U. Comparison of the results of short-term rigid and semi-rigid cast immobilization for the treatment of grade 3 inversion injuries of the ankle. *Injury* 1998;**29** (8):581–4.

Brakenbury 1983 {published and unpublished data} Brakenbury PH, Kotowski J. A comparative study of the management of ankle sprains. British Journal of Clinical Practice 1983;37(5):181–5.

Brooks 1981 *{published data only (unpublished sought but not used)}* Brooks SC, Potter BT, Rainey JB. Treatment for partial tears of the lateral ligament of the ankle: a prospective trial. *BMJ* 1981;**282**:606–7.

Brostrom 1966 {published data only (unpublished sought but not used)}

Brostrom L. Sprained ankles V. Treatment and prognosis in recent ligament ruptures. *Acta Chirurgica Scandinavica* 1966;**132**(5):537–50. [MEDLINE: 67182970]

Caro 1964 {published data only}

Caro D, Craft IL, Howells JB, Shaw PC. Diagnosis and treatment of injury of the lateral ligament of the ankle joint. *Lancet* 1964;**171**:720–3.

Cetti 1984 {published data only}

* Cetti R, Christensen SE, Corfitzen MT. Ruptured fibular ankle ligament: plaster or Pliton brace ?. *British Journal of Sports Medicine* 1984;**18**(2):104–9. Cetti R, Frederiksen E, Christensen SE, Kjaersgaard E, Lund H, Corfitsen MT. Rupture of the fibular ankle ligaments treated with plaster immobilization or mobile Pliton-80 bandage. A prospective randomized study. *Ugeskrift for Laeger* 1985;**147**(11):941–944.

Dettori 1994 {published and unpublished data}

Dettori JR, Basmania CJ. Early ankle mobilization, Part II: A one-year follow-up of acute, lateral ankle sprains (a randomized clinical trial). *Military Medicine* 1994;**159**(1): 20–4. [MEDLINE: 94217940]

* Dettori JR, Pearson BD, Basmania CJ, Lednar WM. Early ankle mobilization, Part I: The immediate effect on acute, lateral ankle sprains (a randomized clinical trial). *Military Medicine* 1994;**159**(1):15–20. [MEDLINE: 94217939]

Eiff 1994 {published and unpublished data}

Eiff MP, Smith AT, Smith GE. Early mobilization versus immobilization in the treatment of lateral ankle sprains. *American Journal of Sports Medicine* 1994;**22**(1):83–8.

Freeman 1965a {published data only (unpublished sought but not used)}

Freeman MAR. Treatment of ruptures of the lateral ligament of the ankle. *Journal of Bone and Joint Surgery. British Volume* 1965;**47**(4):661–8.

Gronmark 1978 {published data only (unpublished sought but not used)}

Gronmark T, Johnsen O, Kogstad O. Rupture of the lateral ligaments of the ankle: a controlled clinical trial. *Injury* 1980;**11**:215–8.

Hedges 1980 {published data only (unpublished sought but not used)}

Hedges JR, Anwar RA. Management of ankle sprains. Annals of Emergency Medicine 1980;**9**(6):27–31. [MEDLINE: 80218382]

Klein 1991 {published data only}

Klein J, Rixen D, Albring T, Tiling T. Functional versus plaster cast treatment of acute rupture of the fibular ligament of the upper ankle joint. A randomized clinical study [Funktionelle versus Gipsbehandlung bei der frischen Aussenbandruptur des oberen Sprunggelenks. Eine randomisierte klinische Studie]. *Unfallchirurg* 1991;**94**(2): 99–104. [MEDLINE: 91240295]

Konradsen 1991 {published data only (unpublished sought but not used)}

Konradsen L, Holmer P, Sondergaard L. Early mobilizing treatment for grade III ankle ligament injuries. *Foot & Ankle* 1991;**12**(2):69–73. [MEDLINE: 92128821]

Korkala 1987 {published and unpublished data}

Korkala O, Rusanen M, Jokipii P, Kytomaa J, Avikainen V. A prospective study of the treatment of severe tears of the lateral ligament of the ankle. *International Orthopaedics* 1987;**11**(1):13–7.

Lind 1984 {published data only}

Lind T. Conservative treatment of rupture of the lateral ligament of the ankle [Konservativ behandling af laterale ligamentrupturer i fodleddet]. *Ugeskrift for Laeger* 1984; **146**(51):4017–9.

Milford 1990 {published and unpublished data}

Milford PI, Dunleavy PJ. A pilot trial of treatment of acute inversion sprains to the ankle by ankle supports. *Journal of the Royal Naval Medical Service* 1990;**76**(2):97–100.

Moller-Larsen 1988 {published and unpublished data}

Moller-Larsen F, Wethelund JO, Jurik AG, de Carvalho A, Lucht U. Comparison of three different treatments for ruptured lateral ankle ligaments. *Acta Orthopaedica Scandinavica* 1988;**59**(5):564–6.

Munk 1995 {published data only}

Munk B, Holm-Cristensen K, Lind T. Long-term outcome after ruptured lateral ankle ligaments. *Acta Orthopaedica Scandinavica* 1995;**66**(5):452–454.

Regis 1995 {published data only}

Regis D. Dynamic Orthopaedic brace in the treatment of ankle sprains. *Foot and Ankle International* 1995;**16**(7): 422–6.

Roycroft 1983 {published and unpublished data}

Roycroft S, Mantgani AB. Treatment of inversion injuries of the ankle by early active management. *Physiotherapy* 1983; **69**(10):355–6.

Sommer 1993 {published and unpublished data}

Sommer HM, Schreiber H. Early functional conservative therapy of a fresh fibular rupture of the capsular ligament from a socioeconomical viewpoint [Die fruh–funktionelle koservative therapie der frishen fibularen kapsel–band–ruptur aus sozial–okonomischer sicht]. *Sportverletzung Sportschaden* 1993;7(1):40–6.

References to studies excluded from this review

Allen 1985 {published data only}

Allen MJ, McShane M. Inversion injuries to the lateral ligament of the ankle joint. A pilot study of treatment. *British Journal of Clinical Practice* 1985;**39**(7):282–5.

Andersson 1983 {published data only}

Andersson S, Fredin H, Sanzen L, Westlin N. Ibuprofen and compression bandage in the treatment of ankle sprains. *Acta Orthopaedica Scandinavica* 1983;**54**:322–5.

Freeman 1965 {published data only}

Freeman MAR. The etiology and prevention of functional instability of the foot. *Journal of Bone and Joint Surgery. British Volume* 1965;**47**(4):678–85.

Holmer 1991 {published data only}

Holmer P, Carstensen NC, Merrild UB. Support stockings versus bandages in the treatment of acute ankle sprains. A prospective random study [Stottestromper kontra stottebind i behandlingen af akutte ankeldistorsioner. En prospektiv randomiseret undersogelse]. *Ugeskrift for Laeger* 1991;**153**: 430–2.

Johannes 1993 {published data only}

Johannes EJ, Sukul DM, Spruit PJ, Putters JL. Controlled trial of a semi-rigid bandage ('Scotchrap') in patients with ankle ligament lesions. *Current Medical Research and Opinion* 1993;**13**(3):154–62. [MEDLINE: 94038093]

Jorgensen 1986 {published data only}

Jorgensen FR, Gotzche PC, Jensen CM, Nielsen BM, Ronholt EB, Tranberg FH. Naproxen and mobilisation in the treatment of acute ankle sprains. *Ugeskrift for Laeger* 1986;**148**(2):1266–8.

Karlsson 1996 {published data only}

Karlsson J, Eriksson BI, Sward L. Early functional treatment for acute ligament injuries of the ankle joint. *Scandinavian Journal of Medicine & Science in Sports* 1996;**6**(6):341–5.

Leanderson 1999 {published data only}

Leanderson J, Bergqvist M, Rolf C, Westblad P, Wigelius-Roovers S, Wredmark T. Early influence of an ankle sprain on objective measures of ankle joint function. *Knee Surgery, Sports Traumatology, Arthroscopy* 1999;7:51–8.

Makuloluwe 1977 {published data only}

Makuloluwe RTB, Mouzas GL. Ultrasound in the treatment of sprained ankles. *Practitioner* 1977;**218**:586–8.

Muwanga 1986 {published data only}

Muwanga CL, Quinton DN, Sloan JP, Gillies P, Dove AF. A new treatment of stable lateral ligament injuries of the ankle joint. *Injury* 1986;**17**:380–2.

Nilsson 1983 {published data only}

Nilsson S. Sprains of the lateral ankle ligaments. *Journal of the Oslo City Hospitals* 1983;**33**:13–36.

Oostendorp 1987 {published data only}

Oostendorp RAB. The functional instability after an inversion trauma of ankle and foot [Functionele instabiliteit na het inversietrauma van enkel en voet: een effectonderzoek

pleisterbandage versus pleisterbandage gecombineerd met fysiotherapie]. *Geneeskunde en sport* 1987;**20**(2):45–55.

Otto 1997 {published data only}

Otto M, Novak L, Fekecs G. Functional conservative versus operative treatment of outer ankle ligament-ruptures. Comparative study [abstract]. *Journal of Bone and Joint Surgery. British Volume* 1997;**79 Suppl 2**:250.

Scotece 1992 {published data only}

Scotece GG, Guthrie MR. Comparison of three treatment approaches for grade I and II ankle sprains in active duty soldiers. *Journal of Orthopaedic & Sports Physical Therapy* 1992;**15**(1):19–22.

Twellaar 1993 {published data only}

Twellaar M, Veldhuizen JW, Verstappen FT. Ankle sprains. Comparison of long-term results of functional treatment methods with adhesive tape and bandage ("brace") and stability measurement [Das Knochelinversionstrauma. Vergleich der Langzeitergebnisse funktioneller Behandlungsmethoden mit Klebeverband und Bandage ("Brace") und die Eignung der Stabilometrie]. *Unfallchirurg* 1993;**96**(9):477–82. [MEDLINE: 94053819]

Viljakka 1983 {published data only}

Viljakka T, Rokkanen P. The treatment of ankle sprain by bandaging and antiphlogistic drugs. *Annales Chirurgiae et Gynaecologiae* 1983;**72**:66–70.

Wilkerson 1993 {published data only}

Wilkerson GB, Horn-Kingery HM. Treatment of the inversion ankle sprain: comparison of different modes of compression and cryotherapy. *Journal of Orthopaedic & Sports Physical Therapy* 1993;**17**(5):240–5.

References to studies awaiting assessment

Duwairi 1998 {published data only}

Duwairi MQ. Functional treatment of acute ankle sprain. *Saudi Medical Journal* 1998;**19**(3):329–331.

Grasmueck 1997 {published data only}

Grasmueck J, Lohrer W, Alt, W. A prospective and randomized study concerning operative and conservative treatment of acute ankle sprains in sportsmen/women [abstract]. *Orthopaedic Transactions* 1997;**21**(4):1229–30.

Hoogenband 1984 {published data only}

Hoogenband van den CR, Moppes van FI, Coumans PF, Stapert JWJL, Greep JM. Study on clinical diagnosis and treatment of lateral ligament lesion of the ankle joint. A prospective clinical randomized trial. *International Journal of Sports Medicine* 1984;**5**(Suppl):159–61. Hoogenband van den CR, Van Moppes FI, Stapert JW, Greep JM. Clinical diagnosis, arthrography, stress

examination and surgical findings after inversion trauma of the ankle. *Archives of Orthopaedic and Trauma Surgery* 1984; **103**(2):115–119.

Van Moppes FI, Van den Hoogenband CR. *Diagnostic* and therapeutic aspects of inversion trauma of the ankle joint [thesis]. Maastricht: Univ. of Maastricht, 1982.

Rocinski 1991 {published data only}

Rocinski TJ. The effect of intermittent compression on edema in postacute ankle sprains. *Journal of Orthopaedic & Sports Physical Therapy* 1991;14(2):65–9.

Soosai Nathan 1997 {published data only}

Soosai Nathan S, Nwachukwu I, Forester A. A prospective randomised trial comparing the aircast ankle brace with conservative treatment for lateral ligament injuries of the ankle [abstract]. *Journal of Bone and Joint Surgery. British Volume* 1997;**79 Suppl 2**:250.

Vitellas 1995 {published data only}

Vitellas KM, Mueller CF, Blau NA, Verner JJ, Zuelzer WA. The role of stress radiographs for the severe ankle sprain: A 7-year prospective study. *Emergency Radiology* 1995;**2**(6): 339–344. [: EMBASE 1996027291]

Zwipp 1986 {published data only}

Zwipp H, Hoffmann R, Thermann H, Wippermann BW. Rupture of the ankle ligaments. *International Orthopaedics* 1991;**15**:245–9.

Zwipp H, Hoffmann R, Wippermann B, Thermann H, Gottschalk F. Rupture of the fibular ligament of the upper ankle joint. *Orthopade* 1989;**18**(4):336–341.

* Zwipp H, Tscherne H, Hoffmann R, Wippermann B. Therapy of fresh fibular ligament ruptures. *Orthopade* 1986; 15(6):446–453.

Zwipp H, Tscherne H, Hoffman R, Thermann H. [Rib der knochelbander: operativ oder konservativ?]. *Deutsche Arzteblatt* 1988;**42**:2019.

Additional references

Alder 1976

Adler H. Therapy and prognosis of fresh external ankle ligament lesions [Therapie und prognose der frischen aussenknochelbandlesion]. *Unfallheilkunde* 1976;**79**(3): 101–4. [MEDLINE: 76272159]

Bernett 1979

Bernett P, Schirmann A. Acute sporting injuries of the ankle joint [Sportverletzungen des sprunggelenkes]. *Unfallheilkunde* 1979;**82**(4):155–60. [MEDLINE: 79204615]

Brostrom 1965

Brostrom L. Sprained ankles III. Clinical observations in recent ligament ruptures. *Acta Chirurgica Scandinavica* 1965;**130**:560–9.

Brostrom 1966a

Brostrom L. Sprained ankles. VI. Surgical treatment of chronic ligament ruptures. *Acta Chirurgica Scandinavica* 1966;**132**:551–65.

Clarke 2001

Clarke M, Oxman AD, editors. Optimal search strategy for RCTs. Cochrane Reviewers Handbook 4.1.4 [updated October 2001], Appendix 5c. In: The Cochrane Library, Issue 4, 2001. Oxford: Update Software. Updated quarterly.

Dehne 1933

Dehne E. [Die klinik der frischen und habituellen adduktion–supinations distorsion des fusses]. *Deutsche Zeitschrift fur Chirurgie* 1933;**242**:40–61.

Dickersin 1992

Dickersin K, Berlin JA. Meta-analysis: state-of-the-science. *Epidemiologic Reviews* 1992;**14**:154–76.

Freeman 1965b

Freeman MAR. The etiology and prevention of functional instability of the foot. *Journal of Bone and Joint Surgery. British Volume* 1965;**47**:678–85.

Freeman 1965c

Freeman MAR. Instability of the foot after injuries to the lateral ligament of the ankle. *Journal of Bone and Joint Surgery. British Volume* 1965;**47**:669–77.

Jacob 1986

Jacob RP, Raemy H, Steffen R, Zeegers AVCM. [Zur funktionellen behandlung des frischen aussenbaderrisses mit der Aircast–Schiene]. *Orthopade* 1986;**14**:434–440.

Jadad 1996

Jadad AR, Moore A, Carroll D, Jenkinson DJM, Reynolds DJM, Gavaghan DJ, et al.Assessing the quality of reports of randomized clinical trials: is blinding necessary?. *Controlled Clinical Trials* 1996;17:1–12.

Kannus 1991

Kannus P, Renstrom P. Current concept review. Treatment for acute tears of the lateral ligaments of the ankle. *Journal of Bone and Joint Surgery. American Volume* 1991;73:305–12.

Kannus 1991b

Kannus P, Renstrom P, Jarvinen M. Acute rupture of ankle ligaments - operation, immobilization or functional treatment?. *Duodecim* 1991;**107**(1):15–24.

Karlsson 1997

Karlsson J, Eriksson BI, Bergsten T, Rudholm O, Sward L. Comparison of two anatomic reconstructions for chronic lateral instability of the ankle joint. *American Journal of Sports Medicine* 1997;**25**(1):48–53.

Katcherian 1994

Katcherian DA. Soft-tissue injuries of the ankle. In: Lutter LD, Mizel MS, Pfeffer GB editor(s). *Orthopaedic Knowledge Update: Foot and Ankle*. Rosemont, Illinois, 1994:241–53.

Keeman 1990

Keeman JN. [Commentaar Enkelspecial]. *Reuma en Trauma* 1990;**1**:34–5.

Kerkhoffs 2002a

Kerkhoffs GMMJ, Struijs PAA, Marti RK, Assendelft WJJ, Blankevoort L, Dijk van CN. Different functional treatment strategies for acute lateral ankle ligament injuries in adults (Cochrane review). *The Cochrane Library* 2002, Issue 3.

Kerkhoffs 2002b

Kerkhoffs GMMJ, Handoll HHG, Rowe BH, de Bie R, Struijs PAA. Surgical versus conservative treatment for acute injuries of the lateral ligament complex of the ankle in adults (Cochrane review). *The Cochrane Library* 2002, Issue 3.

Lassiter 1989

Lassiter TE, Malone TR, Garret WE. Injuries to the lateral ligaments of the ankle. *Orthopedic Clinics of North America* 1989;**20**:629–40.

Leonard 1949

Leonard MH. Injuries of the lateral ligaments of the ankle. A clinical and experimental study. *Journal of Bone and Joint Surgery. American Volume* 1949;**31**:373–7.

Lindenfeld 1994

Lindenfeld TN, Schmitt DJ, Hendy MP, Mangine RE, Noyes FR. Incidence of injury in indoor soccer. *American Journal of Sports Medicine* 1994;**22**(3):364–71.

Luidinga 1985

Luidinga F, Rogmans WHJ. [Epidemiologie van acute sportletsels]. *Nederlands Tijdschrift voor Geneeskunde* 1985; **129**:1051–4.

Marti 1982

Marti RK. [Bagatelletsels van de voet]. *Capita Selecta, Reuma Wereldwijd* 1982:56.

Ogilvie-Harris 1995

Ogilvie-Harris DJ, Gilbart M. Treatment modalities for soft tissue injuries of the ankle: a critical review. *Clinical Journal* of Sport Medicine 1995;**5**(3):175–86.

Schulz 1994

Schulz KF, Chalmers I, Grimes DA, Altman DG. Assessing the quality of randomization from reports of controlled trials published in obstetrics and gynaecology journals. *JAMA* 1994;**272**:125–8.

Shrier 1995

Shrier I. Treatment of lateral collateral ligament spains of the ankle: a critical appraisal of the literature. *Clinical Journal of Sport Medicine* 1995;**5**(3):187–95.

Stover 1980

Stover CN. Air stirrup management of ankle injuries in the athlete. *American Journal of Sports Medicine* 1980;8: 360–365.

Tiling 1994

Tiling T, Bonk A, Hoher J, Klein J. Acute injury to the lateral ligament of the ankle joint in the athlete [Die akute Aussenbandverletzung des Sprunggelenks beim Sportler]. *Chirurg* 1994;**65**(11):920–33. [MEDLINE: 95121054]

Vaes 1985

Vaes P, de Boeck H, Handelberg F, Oxman AD. Comparative radiologic study of the infuence of ankle joint bandages on ankle stability. *American Journal of Sports Medicine* 1985;**13**:46–50.

van Dijk 1994

van Dijk CN. On diagnostic strategies in patients with severe ankle sprain [thesis]. Amsterdam, the Netherlands: Univ. of Amsterdam, 1994.

van Moppes 1982

van Moppes FI, van den Hoogenband CR. Diagnostic and therapeutic aspects of inversion trauma of the ankle joint

[thesis]. Maastricht, the Netherlands: Univ. of Maastricht, 1982.

Verhagen 1998

Verhagen AP, de Vet HCW, de Bie RA, Kessels AGH, Boers M, Knipschild PG. Balneotherapy and quality assessment: interobserver reliability of the Maastricht criteria list and the need for blinded quality assessment. *Journal of Clinical Epidemiology* 1998;**51**(4):335–341.

Watson-Jones 1976

Watson-Jones R. Fractures and joint injuries. 5th Edition. Edinburgh: Churchill Livingstone, 1976.

Wiersma 1998

Wiersma PH. Dynamic loading of the lateral ankle

ligament complex: an experimental, clinical and anatomical study [thesis]. Amsterdam, The Netherlands: Univ. of Amsterdam, 1998.

Zeegers 1995

Zeegers AVCM. *Het supinatieletsel van de enkel*. Utrecht, the Netherlands: Univ. of Utrecht, 1995.

References to other published versions of this review

Kerkhoffs 2001

Kerkhoffs GMMJ, Rowe B, Assendelft WJJ, Kelly KD, Struijs PAA, van Dijk CN. Immobilisation for acute ankle sprain. a systematic review. *Archives of Orthopaedic and Trauma Surgery* 2001;**121**:462–71.

* Indicates the major publication for the study

CHARACTERISTICS OF STUDIES

Characteristics of included studies [ordered by study ID]

Avci 1998

Methods	Randomisation method: Last digit on patients' chart number. Blinding: no blinding. Loss to FU: 7/64 (11%) A: 0 E: 0 J: 2 B: 1 F: 0 K: 1 C: 0 G: 2 L:2(0) D: 1 H:2 T: 11(9)		
Participants		University Hospital, Ankara, Turkey. Skeletally mature patients presenting with an acute grade 3 inversion injury of the ankle. Age range 17-47 years, 65% male. Exclusion criteria: fractures, history of chronic instability	
Interventions	Period of study: not stated a) Rigid cast for 2 weeks. b) Semi-rigid cast for 2 weeks. Assigned: 32/32 Analysed: 26/31		
Outcomes	Return to work Pain (Dichotomous) Swelling Objective instability ROM		
Notes	Diagnosis, Phys. exam.		
Risk of bias			
Item	Authors' judgement	Description	
Allocation concealment?	No	C - Inadequate	
Brakenbury 1983			
Methods	Randomisation method: not stated. Blinding: no blinding. Loss to FU: 47%, at 14d. in elastic support group, 22% in cast group A:1 E:0 J:2 B:1 F:0 K:1 C:0 G:1 L:0 D:2 H:2 T:10		
Participants	Middlesbrough General Hospital, Accident and Emergency department, UK. Men 20-40 years presenting with an acute ankle sprain, (% med. lig. sprain <10%) Excusion criteria: fractures or other pathology of		

Brakenbury 1983 (Continued)

	ankle joint; other injuries in the same limb; patient taking medication	
Interventions	Period of study: not mentioned. a) Cast + enzyme/placebo for 1 week b) Tubigrips + enzyme/placebo for 1 week Assigned: 200/200 Analysed: 106/156	
Outcomes	Return to work Swelling ROM	
Notes	Diagnosis, Phys. exam X-ray	
Risk of bias		
Item	Authors' judgement	Description
Allocation concealment?	Unclear	B - Unclear
Brooks 1981		
Methods	Randomisation method: not mentioned Blinding: no blinding. Loss to FU: 63/165 (38%) A:2 E:0 J:2 B:1 F:0 K:1 C:1(0)G:2 L:1(0) D:2 H:2 T:14(12)	
Participants	Regional accident unit, Edinburgh, Scotland. Skeletally mature individuals presenting with an acute ankle sprain. Exclusion criteria: Age <12, > 65. Fractures.	
Interventions	Period of study: not stated a) Cast for period of complaints Assigned: ? Analysed: 26 b) Physiotherapy for period of complaints Assigned: ? Analysed: 21	

Immobilisation and functional treatment for acute lateral ankle ligament injuries in adults (Review) Copyright © 2009 The Cochrane Collaboration. Published by John Wiley & Sons, Ltd.

d) No treatment or only a minimal bandage

c) Double tubigrip Assigned: ? Analysed: 28

Brooks 1981 (Continued)

	Assigned: 27 Analysed: 26/21		
Outcomes	Return to work Satisfaction		
Notes	Diagnosis, Stress X-ray Phys. exam. Due to poor reporting of results, little data could be contributed to the analysis		
Risk of bias			
Item	Authors' judgement	Description	
Allocation concealment?	Unclear	B - Unclear	
Brostrom 1966			
Methods	Randomisation method: odd/even patient numbers. Blinding: no blinding Loss to FU: 2/188, at mean 3.8 years. A:0 E:0 J:2 B:1 F:0 K:2 C:0 G:2 L:1 D: 2 H:1 T: 11		
Participants	Karolinska Hospital, Stockholm, Sweden. Skeletally mature individuals presenting with recent ligament rupture. 76% male, age range not stated. Exclusion criteria: fractures		
Interventions	Period of study: 1961-1962 a) Cast for 2 weeks b) Elastic bandage for 2 weeks Assigned: 83/105 Analysed: 82/104		
Outcomes	Return to work Subjective instability		
Notes	Diagnosis, Arthrography		
Risk of bias	Risk of bias		
Item	Authors' judgement	Description	
Allocation concealment?	No	C - Inadequate	

Caro 1964

Methods	Randomisation method: 'random allocation'. Blinding: no blinding. Loss to FU: 9/88, at 3 months. A:1 E:0 J:1 B:2 F:0 K:2 C:0 G:1 L:2 D:1 H:1 T:11		
Participants	St James' Hospital, London, UK. Individuals aged 20-40 years presenting with an inv	version injury (% fractures <10%)	
Interventions	a) Cast for 2 weeks b) Tape, strapping for 2 weeks Assigned: 41/47 Analysed: 37/42	b) Tape, strapping for 2 weeks Assigned: 41/47	
Outcomes	Return to work		
Notes	Diagnosis, Stress X-ray Phys. exam.		
Risk of bias			
Item	Authors' judgement	Description	
Allocation concealment?	Unclear	B - Unclear	
Cetti 1984			
Methods	Randomisation method: 'assigned randomly.' Blind A:1 E:0 J:2 B:0 F:0 K:1 C:0 G:0 L:1 D:0 H:1 T: 6	ling: no blinding. Loss to FU: 0%, at 6 months.	
Participants	Frederiksberg University Hospital, Kopenhagen, Denmark. Skeletally mature individuals presenting with an acute ankle sprain. Age range 13-76 yrs, 53% male. Exclusion criteria not stated		
Interventions	Period of study: 1981-1982 a) Cast for 6 weeks b) Brace for 6 weeks Assigned: 65/65 Analysed: 65/65		
Outcomes	Return to sports Return to work Pain (Dichotomous) Swelling Objective instability		

Cetti 1984 (Continued)

	ROM Recurrent sprain		
Notes	Diagnosis, Stress X-ray Phys.exam.		
Risk of bias			
Item	Authors' judgement	Description	
Allocation concealment?	Unclear	B - Unclear	
Dettori 1994			
Methods	Randomisation method: not stated. Blinding: adequ at 1 year. A:1 E:0 J:2 B:0 F:0 K:1 C:0 G:2 L:0 D:1 H:2 T:9	ate blinding of outcome measures. Loss to FU: 8/70,	
Participants	Madigan Army Medical Centre, Tacoma, USA. Active duty military members presenting with an acute ankle sprain. Exclusion criteria: fractures; injury older than 72 hours; prior ankle sprain within 6 months		
Interventions	Period of study: 1991-1992 a) Cast for 2 weeks b) Brace for 2 weeks Assigned: 22/48 Analysed: 16/46		
Outcomes	Return to work Pain (dichotomous) Swelling Recurrent sprain		
Notes	Diagnosis, Stress X-ray Phys.exam.		
Risk of bias	Risk of bias		
Item	Authors' judgement	Description	
Allocation concealment?	Unclear	B - Unclear	

Eiff 1994

Methods	Randomisation method: block randomisation. Blinding: no blinding. Loss to FU: 5/82, at 1 year. A:2(1) E:0 J:0 B:1 F:0 K:2 C:0 G:2 L:1 D:2 H:2 T:12(11)	
Participants	Oregon, University Hospital, Oregon, USA. Active duty military members, military dependants, retirees presenting with a lateral ankle sprains. Exlusion criteria: time from injury to presentation > 48 hours; residual symptoms prior ankle injury; no swelling or ligamentous tenderness at presentation; ankle instability as measured by stress radiographs was present	
Interventions	Period of study: 1989-1990 a) Sugar-tong plaster splint and crutches for 10 days b) Elastic wrap for 2 days, followed by brace (Air-Stirrup) for 1 week. Assigned: 41/41 Analysed: 37/40	
Outcomes	Return to sports Return to work Pain (dichotomous) Swelling Subjective instability Recurrent sprain	
Notes	Diagnosis, Phys.exam. Stress X-ray	
Risk of bias		
Item	Authors' judgement	Description
Allocation concealment?	Yes	A - Adequate
Freeman 1965a		
Methods	Randomisation method: 'randomly selected'. Blinding: no blinding. Loss to FU: 0/28, at 1 year. A:1 E:0 J:2 B:1(0) F:0 K:1 C:0 G:1(0) L:1 D:0 H:2 T: 9(7)	
Participants	Middlesex Hospital, London, UK. Skeletally mature individuals (mainly young men) presenting with rupture of the lateral ligament of the ankle. Exclusion criteria: fractures	
Interventions	Period of study: 1963 a) Cast for 6 weeks	

Freeman 1965a (Continued)

	b) Strapping and mobilisation for period of complaints Assigned: 16/12 Analysed: 16/12		
Outcomes	Return to sports Pain (dichotomous) Swelling Subjective instability Objective instability		
Notes	Diagnosis, Phys.exam. X-ray		
Risk of bias			
Item	Authors' judgement		Description
Allocation concealment?	Unclear		B - Unclear
Gronmark 1978			
Methods	Randomisation method: not stated. Blinding: no blinding. Loss to FU: 0%, at 17 months. A:1 E:0 J:1 B:2 F: 0 K:2 C:0 G:2 L:2 D:0 H:1 T: 11		
Participants	Telemark Central Hospital, Skien, Norway. Skeletally mature individuals presenting with ruptured lateral ankle ligaments. Age range 14-53 years, 70% male. Exclusion criteria: fractures		
Interventions	Period of study: 1976-1977 a) Cast for 2 weeks, walking cast for 4 weeks b) Wrap for 6 weeks Assigned: 33/30 Analysed: 33/30		
Outcomes	Pain (dichotomous)		
Notes	Diagnosis, Stress X-ray Phys.exam.		
Risk of bias			
Item	Authors' judgement		Description

Gronmark 1978 (Continued)

Allocation concealment?	Unclear	B - Unclear
Hedges 1980		
Methods	Randomisation method: by chart number. Blinding: no blinding. Loss to FU: 18% at 1 year. A:0 E:0 J:2 B:0 F:0 K:2(1) C:0 G:2(1) L:1 D:2 H:2 T:11(9)	
Participants	Medical College of Pennsylvania Hospital, Philadelphia, Pennsylvania, USA. Skeletally mature individuals (>15, <65 years) presenting with ankle injury. Exclusion criteria: fractures	
Interventions	Period of study: 1976-1977 a) Cast for period of complaints b) Wrap for period of complaints Assigned: 61/60 Analysed: 49/44	
Outcomes	Pain (not mentioned) Swelling	
Notes	Diagnosis, Phys. exam. X-ray	
Risk of bias		
Item	Authors' judgement	Description
Allocation concealment?	No	C - Inadequate
Klein 1991		
Methods	Randomisation method: not described. Blinding: no blinding. Loss to FU: 10% at 15 months. A:0 E:0 J:2 B:1 F:0 K:1 C:0 G:1 L:2 D:2 H:2 T:11	
Participants	University Hospital of Cologne. Cologne, Germany. Skeletally mature basketball players presenting with recent rupture of the fibular ligament of the ankle. Exclusion criteria: fractures	
Interventions	Period of study: 1989-1990 a) Cast for 6 weeks b) Brace (Aircast) for 6 weeks Assigned: 39/89	

Klein 1991 (Continued)

	Analysed: 39/89	
Outcomes	Return to sports Pain (dichotomous) Swelling Subjective instability Recurrent sprain Patient satisfaction	
Notes	Diagnosis, Stress X-ray Phys. exam.	
Risk of bias		
Item	Authors' judgement	Description
Allocation concealment?	Unclear	B - Unclear

Konradsen 1991

Methods	Randomisation method: not described. Blinding: no blinding. Loss to FU: 7/80, at 1 year. A:1 E:0 J:2 B:0 F:0 K:2 C:0 G:1 L:1 D:0 H:2 T:9
Participants	Horsholm Hospital, Horsholm, Denmark. Adults presenting with grade III ruptures of lateral ankle ligaments. Exclusion criteria: sprain more than 24 hours old; fractures; history of sprains; no other acute or chronic immobilising lesion
Interventions	Period of study: not described a) Cast for 1 week followed by a walking cast for 5 weeks b) Brace (Aircast) for 6 weeks Assigned: 40/40 Analysed: 37/36
Outcomes	Return to work Return to sports
Notes	Diagnosis, Stress X-ray Phys.exam.
Risk of bias	

Konradsen 1991 (Continued)

Item	Authors' judgement	Description		
Allocation concealment?	Unclear	B - Unclear		
Korkala 1987				
Methods	Randomisation method: randomly divided by 'close at 2 years. A:2(0) E:0 J:2 B:0 F:0 K:2 C:0 G:2(0) L:1 D:1(0) H:2 T:12(7)	d letters'. Blinding: no blinding. Loss to FU: 33/150,		
Participants	University Central Hospital, Helsinki, Finland. Skeletally mature patients (>15, <50 y) presenting with severe ankle sprains. Exclusion criteria: alcoholism; serious medical conditions; psychiatric illness; fractures			
Interventions	Period of study: not described a) Cast for 4 weeks b) Semi-elastic bandage (Tenoplast) for 1 week, foll- Assigned: 50/50 Analysed: 47/36	owed by elastic bandage for 1-3 weeks.		
Outcomes	Return to sports Subjective instability Objective instability Recurrent sprain			
Notes	Diagnosis, Phys.exam. X-ray			
Risk of bias				
Item	Authors' judgement	Description		
Allocation concealment?	Yes A - Adequate			

Lind 1984

Methods	Randomisation method: drawing of lots. Blinding: no blinding Loss to FU: 10/66, at six months. A:1 E:0 J:2(1) B:1 F:0 K:1 C:0 G:2(0) L:1 D:0 H:2 T:10(7)				
Participants	Fra Haderslev Sygehus, Haderslev, Denmark. Skel ligament rupture. Exclusion criteria: fractures; injur	etally mature patients presenting with lateral ankle y more than 24 hours old			
Interventions	Period of study: 1983 a) Cast for 5 weeks b) Bandage (Pronating, supportive) for period of complaints Assigned: 33/33 Analysed: 29/27				
Outcomes	Pain (dichotomous) Objective instability Recurrent sprain				
Notes	Diagnosis, Arthrography				
Risk of bias					
Item	Authors' judgement	Description			
Allocation concealment?	Unclear	B - Unclear			
Milford 1990					
Methods	Randomisation method: alternate patients. Blinding: no blinding. Loss to FU: 0%, at 3 months. A:0 E:0 J:2 B:0 F:0 K:0				
	C:0 G:1 L:0 D:1 H:2 T:6				
Participants	D:1 H:2 T:6	arine recruits presenting with acute ankle inversion nt injury to other soft tissue structures			

Milford 1990 (Continued)

Outcomes	Return to sports			
Notes	Diagnosis, Stress X-ray Phys. exam.			
Risk of bias				
Item	Authors' judgement	Description		

Moller-Larsen 1988

Methods	Randomisation method: concealed envelope drawing. Blinding: no blinding. Loss to FU: 10/130, at 1 year. A:2(0) E:0 J:0 B:0 F:0 K:1 C:0 G:2 L:1 D:2(0) H:2 T:10(6)				
Participants	Arhus Municipal Hospital, Arhus, Denmark. Skeletally mature individuals (>15, <50y) presenting v fractures	with lateral ankle ligament rupture. Exclusion criteria:			
Interventions	Period of study: 1983- 1985 a) Cast for 5 weeks b) Tape for 5 weeks Assigned: 60/70 Analysed: 55/65				
Outcomes	Return to sports Swelling Satisfaction				
Notes	Diagnosis, Arthrography				
Risk of bias	Risk of bias				
Item	Authors' judgement	Description			
Allocation concealment?	Yes	A - Adequate			

Munk 1995

Methods	Randomisation method: concealed envelope drawing. Blinding: no blinding. Loss to FU: 48/108 at 11 years. A: 2(1) E:0 J:2 B: 1 F:0 K:2 C: 0 G:0 L:2 D:2 H:2 T:13(12)			
Participants	Haderslev Hospital, Haderslev, Denmark. Skeletally ankle ligaments. Exclusion criteria: injury more that	mature patients presenting with acute ruptured lateral 1 24h; fractures		
Interventions	Period of study: 1979-1982 a) Cast for 5 weeks b) Wrap for 5 weeks Assigned: 75/33 Analysed: 44/16			
Outcomes	Pain Recurrent sprain Objective instability			
Notes	Diagnosis, Arthrography			
Risk of bias				
Item	Authors' judgement	Description		
Allocation concealment?	Yes	A - Adequate		
Regis 1995				
Methods	Randomisation method: concealed envelope drawin of 22 months. A:2(1) E:0 J: 2(0) B:2(0) F:0 K: 2(1) C:0 G:1(0) L:2(1) D:2 H:2 T:15(7)	g. Blinding: no blinding. Loss to FU: 0% at a mean		
Participants	Ospedale Policlinico, Verona, Italy. Skeletally mature ankle sprains. Exclusion criteria: fractures	e patients (<30 years) presenting with acute inversion		
Interventions	Period of study: 1993 a) Immobilisation gutter for 3-5 days followed by a b) Immobilisation gutter for 3-5 days followed by a	weightbearing cast for 25 days weightbearing cast for 10 days and a dynamic brace		

Regis 1995 (Continued)

Outcomes	Return to sports Recurrent sprain		
Notes	Diagnosis, Stress X-ray Phys. exam.		
Risk of bias			
Item	Authors' judgement	Description	
Allocation concealment?	Yes	A - Adequate	
Roycroft 1983			
Methods	Randomisation method: not stated. Blinding: no bl A:0 E:0 J:2 B:0 F:0 K:1 C:0 G:1 L:2 D:0 H:2 T:8	inding. Loss to FU: 37/80 at 8 to 12 months.	
Participants	General Hospital, Sligo, Ireland. Skeletally mature individuals presenting with Grade I & II ankle sprains. Exclusion criteria: bony abnormalities		
Interventions	Period of study: not mentioned. a) Cast for 2 weeks b) Wrap for 2 weeks Assigned: 37/43 Analysed: 21/22		
Outcomes	Return to sports Satisfaction		
Notes	Diagnosis, Phys.exam X-ray		
Risk of bias			
Item	Authors' judgement	Description	
Allocation concealment?	Unclear B - Unclear		

Sommer 1993

Methods	Randomisation method: not stated. Blinding: no blinding. Loss to FU: 17/124 at 6 weeks and 18/124 at 1 year. A:1 E:0 J:1 B:0 F:0 K:1 C:0 G:0 L:0 D:0 H:2 T:5				
Participants	University Hospital Heidelberg, Heidelberg, Germany. Patients between 18 and 45 years presenting with an acute lateral ligament rupture. Exclusion criteria: previous injury, chronic instability, fractures, competitive sportsmen				
Interventions	Period of study: 1989 a) Cast for 3 weeks, followed by brace for 3 weeks b) Brace or tape for 6 weeks Assigned: 41/83 Analysed:33/73				
Outcomes	Return to sports Objective instability Recurrent sprain				
Notes	Diagnosis, Arthrography Stress X-ray Phys. exam.				
Risk of bias					
Item	Authors' judgement	Description			
Allocation concealment?	Unclear	Unclear B - Unclear			

Methodological assessment scores in brackets are those initially allocated, and which changed once further information was received from authors.

T = total

Characteristics of excluded studies [ordered by study ID]

Study	Reason for exclusion
Allen 1985	RCT. Three treatment groups: no treatment, tape, tape. Immobilisation was not used as an intervention
Andersson 1983	RCT. Immobilisation was not used as an intervention, all groups received a type of functional treatment

(Continued)

Freeman 1965	RCT. Patients with functional instability of the foot were also studied, therefore the population showed too much heterogeneity. No separate analysis was made for the results of patients with acute ligament injury of the ankle
Holmer 1991	RCT. Immobilisation was not used as an intervention, all groups received a type of functional treatment
Johannes 1993	RCT. Immobilisation was not used as an intervention, all groups received a type of functional treatment
Jorgensen 1986	RCT. Comparison of Naprosyne and mobilisation, and placebo and immobilisation as treatment strategies
Karlsson 1996	RCT. Immobilisation was not used as an intervention, all groups received a type of functional treatment
Leanderson 1999	RCT. Immobilisation was not used as an intervention, all groups received a type of functional treatment. Results were published earlier in 1995 (Leanderson 1995)
Makuloluwe 1977	CCT. The type of immobilisation described does not resemble our definition of immobilisation. It resembles a functional treatment strategy
Muwanga 1986	RCT. Immobilisation was not used as an intervention, all groups received a type of functional treatment
Nilsson 1983	RCT. All treatment groups received an elastic wrap and therefore there was no immobilisation group. Only the effect of functional treatment and physiotherapy was described
Oostendorp 1987	RCT. Immobilisation was not used as an intervention, all groups received a type of functional treatment
Otto 1997	RCT. In this study surgery is compared to a functional treatment
Scotece 1992	RCT. Immobilisation was not used as an intervention, all groups received a type of functional treatment
Twellaar 1993	RCT. Immobilisation was not used as an intervention, all groups received a type of functional treatment
Viljakka 1983	RCT. All treatment groups received a bandage, either elastic or layer, and there was no immobilisation group. Only the effect of functional treatment and antiphlogistic drugs was described
Wilkerson 1993	RCT. Evaluation of different forms of compression and cryotherapy. Immobilisation was not used as an inter- vention

DATA AND ANALYSES

Comparison 1. Immobilisation vs. functional treatment

Outcome or subgroup title	No. of studies	No. of participants	Statistical method	Effect size
1 Numbers not returning to sports	8		Risk Ratio (M-H, Fixed, 95% CI)	Subtotals only
1.1 Short term	3	180	Risk Ratio (M-H, Fixed, 95% CI)	1.01 [0.82, 1.25]
1.2 Intermediate term	4	294	Risk Ratio (M-H, Fixed, 95% CI)	1.32 [0.62, 2.79]
1.3 Long term	5	360	Risk Ratio (M-H, Fixed, 95% CI)	1.86 [1.22, 2.86]
2 Numbers not returning to work	3		Risk Ratio (M-H, Fixed, 95% CI)	Subtotals only
2.1 Short term	2	150	Risk Ratio (M-H, Fixed, 95% CI)	5.75 [1.01, 32.71]
2.2 Intermediate term	3	214	Risk Ratio (M-H, Fixed, 95% CI)	1.14 [0.39, 3.27]
2.3 Long term	1	77	Risk Ratio (M-H, Fixed, 95% CI)	Not estimable
3 Pain	9		Risk Ratio (M-H, Fixed, 95% CI)	Subtotals only
3.1 Short term	3	213	Risk Ratio (M-H, Fixed, 95% CI)	1.42 [0.97, 2.09]
3.2 Intermediate term	5	308	Risk Ratio (M-H, Fixed, 95% CI)	1.28 [0.79, 2.06]
3.3 Long term	5	283	Risk Ratio (M-H, Fixed, 95% CI)	1.17 [0.73, 1.87]
4 Swelling	6		Risk Ratio (M-H, Fixed, 95% CI)	Subtotals only
4.1 Short term	3	260	Risk Ratio (M-H, Fixed, 95% CI)	1.74 [1.17, 2.59]
4.2 Intermediate term	1	77	Risk Ratio (M-H, Fixed, 95% CI)	1.08 [0.07, 16.67]
4.3 Long term	4	280	Risk Ratio (M-H, Fixed, 95% CI)	1.04 [0.57, 1.91]
5 Subjective instability (giving way)	7		Risk Ratio (M-H, Fixed, 95% CI)	Subtotals only
5.1 Short term	1	77	Risk Ratio (M-H, Fixed, 95% CI)	1.51 [0.53, 4.36]
5.2 Intermediate term	3	221	Risk Ratio (M-H, Fixed, 95% CI)	1.24 [0.67, 2.30]
5.3 Long term	5	426	Risk Ratio (M-H, Fixed, 95% CI)	0.91 [0.64, 1.28]
6 Objective instability (talar tilt	5		Risk Ratio (M-H, Fixed, 95% CI)	Subtotals only
and/or ADS)	-			,
6.2 Intermediate term	2	129	Risk Ratio (M-H, Fixed, 95% CI)	0.69 [0.14, 3.34]
6.3 Long term	3	286	Risk Ratio (M-H, Fixed, 95% CI)	0.91 [0.60, 1.39]
7 Recurrent sprain	11		Risk Ratio (M-H, Fixed, 95% CI)	Subtotals only
7.1 Short term	1	77	Risk Ratio (M-H, Fixed, 95% CI)	0.36 [0.04, 3.31]
7.2 Intermediate term	6	456	Risk Ratio (M-H, Fixed, 95% CI)	1.14 [0.69, 1.90]
7.3 Long term	6	487	Risk Ratio (M-H, Fixed, 95% CI)	1.25 [0.84, 1.85]
8 Decreased ROM	2		Risk Ratio (M-H, Fixed, 95% CI)	Subtotals only
8.1 Short term	2	204	Risk Ratio (M-H, Fixed, 95% CI)	1.27 [0.73, 2.19]
8.2 Intermediate term	1	80	Risk Ratio (M-H, Fixed, 95% CI)	7.0 [0.37, 131.28]
9 Patient satisfaction	5	357	Risk Ratio (M-H, Fixed, 95% CI)	1.83 [1.09, 3.07]
9.1 Intermediate term	2	123	Risk Ratio (M-H, Fixed, 95% CI)	4.25 [1.12, 16.09]
9.2 Long term	3	234	Risk Ratio (M-H, Fixed, 95% CI)	1.47 [0.83, 2.60]
10 Return to sports (days)	3	201	Mean Difference (IV, Fixed, 95% CI)	Subtotals only
10.1 Return to sports (days)	3	195	Mean Difference (IV, Fixed, 95% CI)	4.88 [1.50, 8.25]
11 Return to work (days)	6	- / /	Mean Difference (IV, Fixed, 95% CI)	Subtotals only
11.1 Return to work (days)	6	604	Mean Difference (IV, Fixed, 95% CI)	8.23 [6.31, 10.16]
12 Improvement in pain score	1	001	Mean Difference (IV, Fixed, 95% CI)	Totals not selected
12.1 Short term	1		Mean Difference (IV, Fixed, 95% CI)	Not estimable
13 Improvement in swelling	1		Mean Difference (IV, Fixed, 95% CI)	Totals not selected
13.1 Short term	1		Mean Difference (IV, Fixed, 95% CI)	Not estimable

14 Improvement in objective	2	Mean Difference (IV, Random, 95% CI)	Totals not selected
instability (difference in TT)			
14.2 Intermediate term	1	Mean Difference (IV, Random, 95% CI)	Not estimable
14.3 Long term	1	Mean Difference (IV, Random, 95% CI)	Not estimable
15 Improvement in ROM	2	Mean Difference (IV, Fixed, 95% CI)	Totals not selected
15.1 Short term	1	Mean Difference (IV, Fixed, 95% CI)	Not estimable
15.3 Long term	1	Mean Difference (IV, Fixed, 95% CI)	Not estimable

Comparison 2. Immobilisation vs. physiotherapy

Outcome or subgroup title	No. of studies	No. of participants	Statistical method	Effect size
1 Return to work (days)	1		Mean Difference (IV, Fixed, 95% CI)	Totals not selected
1.1 Short term	1		Mean Difference (IV, Fixed, 95% CI)	Not estimable

Comparison 3. Immobilisation vs. other types of immobilisation

Outcome or subgroup title	No. of studies	No. of participants Statistical method		Effect size	
1 Pain	1		Risk Ratio (M-H, Fixed, 95% CI)	Totals not selected	
1.1 Short term	1		Risk Ratio (M-H, Fixed, 95% CI)	Not estimable	
2 Swelling	1		Risk Ratio (M-H, Fixed, 95% CI)	Totals not selected	
2.1 Short term	1		Risk Ratio (M-H, Fixed, 95% CI)	Not estimable	
3 Objective instability (talar tilt and/or ADS)	1		Risk Ratio (M-H, Fixed, 95% CI)	Totals not selected	
3.1 Short term	1		Risk Ratio (M-H, Fixed, 95% CI)	Not estimable	
4 Return to work (days)	1		Mean Difference (IV, Fixed, 95% CI)	Totals not selected	
4.1 Short term	1		Mean Difference (IV, Fixed, 95% CI)	Not estimable	

Comparison 4. Immobilisation vs. functional treatment (high quality trials)

Outcome or subgroup title	No. of studies	No. of participants	Statistical method	Effect size	
1 Numbers not returning to sports	4		Risk Ratio (M-H, Fixed, 95% CI)	Subtotals only	
1.1 Short term	2	107	Risk Ratio (M-H, Fixed, 95% CI)	0.71 [0.31, 1.63]	
1.2 Intermediate term	1	77	Risk Ratio (M-H, Fixed, 95% CI)	3.24 [0.14, 77.06]	
1.3 Long term	4	240	Risk Ratio (M-H, Fixed, 95% CI)	1.70 [0.98, 2.95]	
2 Numbers not returning to work	1		Risk Ratio (M-H, Fixed, 95% CI)	Totals not selected	
2.1 Short term	1		Risk Ratio (M-H, Fixed, 95% CI)	Not estimable	
2.2 Intermediate term	1		Risk Ratio (M-H, Fixed, 95% CI)	Not estimable	
2.3 Long term	1		Risk Ratio (M-H, Fixed, 95% CI)	Not estimable	
3 Pain	5		Risk Ratio (M-H, Fixed, 95% CI)	Subtotals only	

3.1 Short term	1	77	Risk Ratio (M-H, Fixed, 95% CI)	1.31 [0.87, 1.97]
3.2 Intermediate term	2	108	Risk Ratio (M-H, Fixed, 95% CI)	1.17 [0.66, 2.09]
3.3 Long term	4	254	Risk Ratio (M-H, Fixed, 95% CI)	1.23 [0.71, 2.14]
4 Swelling	2		Risk Ratio (M-H, Fixed, 95% CI)	Subtotals only
4.1 Short term	1	77	Risk Ratio (M-H, Fixed, 95% CI)	1.33 [0.75, 2.38]
4.2 Intermediate term	1	77	Risk Ratio (M-H, Fixed, 95% CI)	1.08 [0.07, 16.67]
4.3 Long term	2	131	Risk Ratio (M-H, Fixed, 95% CI)	0.33 [0.01, 7.84]
5 Subjective instability (giving	4		Risk Ratio (M-H, Fixed, 95% CI)	Subtotals only
way)				
5.1 Short term	1	77	Risk Ratio (M-H, Fixed, 95% CI)	1.51 [0.53, 4.36]
5.2 Intermediate term	1	77	Risk Ratio (M-H, Fixed, 95% CI)	1.08 [0.23, 5.03]
5.3 Long term	4	397	Risk Ratio (M-H, Fixed, 95% CI)	0.89 [0.62, 1.30]
6 Objective instability (talar tilt	3		Risk Ratio (M-H, Fixed, 95% CI)	Subtotals only
and/or ADS)				,
6.3 Long term	3	286	Risk Ratio (M-H, Fixed, 95% CI)	0.91 [0.60, 1.39]
7 Recurrent sprain	6		Risk Ratio (M-H, Fixed, 95% CI)	Subtotals only
7.1 Short term	1	77	Risk Ratio (M-H, Fixed, 95% CI)	0.36 [0.04, 3.31]
7.2 Intermediate term	1	77	Risk Ratio (M-H, Fixed, 95% CI)	1.08 [0.23, 5.03]
7.3 Long term	6	487	Risk Ratio (M-H, Fixed, 95% CI)	1.25 [0.84, 1.85]
8 Patient satisfaction	2		Risk Ratio (M-H, Fixed, 95% CI)	Subtotals only
8.3 Long term	2	114	Risk Ratio (M-H, Fixed, 95% CI)	1.68 [0.72, 3.90]
9 Return to work (days)	2		Mean Difference (IV, Fixed, 95% CI)	Subtotals only
9.1 Return to work (days)	2	262	Mean Difference (IV, Fixed, 95% CI)	12.89 [7.10, 18.67]
10 Improvement in pain score	1		Mean Difference (IV, Fixed, 95% CI)	Totals not selected
10.1 Short term	1		Mean Difference (IV, Fixed, 95% CI)	Not estimable
11 Improvement in swelling	1		Mean Difference (IV, Fixed, 95% CI)	Totals not selected
11.1 Short term	1		Mean Difference (IV, Fixed, 95% CI)	Not estimable
12 Improvement in ROM	1		Mean Difference (IV, Fixed, 95% CI)	Totals not selected
12.3 Long term	1		Mean Difference (IV, Fixed, 95% CI)	Not estimable

Analysis I.I. Comparison I Immobilisation vs. functional treatment, Outcome I Numbers not returning to sports.

Review: Immobilisation and functional treatment for acute lateral ankle ligament injuries in adults

Comparison: I Immobilisation vs. functional treatment

Outcome: I Numbers not returning to sports

Study or subgroup	Immobilisation n/N	Functional n/N	Risk Ratio M-H,Fixed,95% Cl	Risk Ratio M-H,Fixed,95% CI
I Short term				,,
Eiff 1994	7/37	7/40	_	1.08 [0.42, 2.79]
Konradsen 1991	37/37	32/36	•	1.12 [0.99, 1.27]
Regis 1995	1/10	8/20	← ∎ ────	0.25 [0.04, 1.73]
Subtotal (95% CI)	84	96	+	1.01 [0.82, 1.25]
Total events: 45 (Immobilisation), Heterogeneity: $Chi^2 = 4.62$, df = Test for overall effect: Z = 0.13 (2 Intermediate term	2 (P = 0.10); $I^2 = 57\%$			
Cetti 1984	6/40	2/40		3.00 [0.64, 3.98]
Dettori 1994	2/18	6/46		0.85 [0.19, 3.84]
Eiff 1994	1/37	0/40		3.24 [0.14, 77.06]
Konradsen 1991	4/37	5/36		0.78 [0.23, 2.67]
Subtotal (95% CI)	132	162		1.32 [0.62, 2.79]
Total events: 13 (Immobilisation), Heterogeneity: $Chi^2 = 2.43$, df = Test for overall effect: $Z = 0.72$ (3 Long term	3 (P = 0.49); $I^2 = 0.0\%$ P = 0.47)			
Eiff 1994	0/37	0/40		0.0 [0.0, 0.0]
Klein 1991	16/26	5/24		2.95 [1.28, 6.82]
Korkala 1987	7/47	8/36		0.67 [0.27, 1.68]
Moller-Larsen 1988	18/55	10/65		2.13 [1.07, 4.22]
Regis 1995	3/10	1/20		6.00 [0.71, 50.59]
Subtotal (95% CI) Total events: 44 (Immobilisation),	· /	185	-	1.86 [1.22, 2.86]
Heterogeneity: $Chi^2 = 7.24$, df = Test for overall effect: Z = 2.86 (
			0.1 0.2 0.5 1 2 5 10	
			Favours immobilis. Favours functional	

Analysis I.2. Comparison I Immobilisation vs. functional treatment, Outcome 2 Numbers not returning to work.

Review: Immobilisation and functional treatment for acute lateral ankle ligament injuries in adults

Comparison: I Immobilisation vs. functional treatment

Outcome: 2 Numbers not returning to work

Study or subgroup	Immobilisation	Functional	Risk Ratio	Risk Ratio
	n/N	n/N	M-H,Fixed,95% Cl	M-H,Fixed,95% CI
I Short term				
Eiff 1994	1/37	1/40		1.08 [0.07, 16.67]
Konradsen 1991	7/37	0/36		4.6 [0.86, 246.68]
Subtotal (95% CI)	74	76		5.75 [1.01, 32.71]
Total events: 8 (Immobilisation),	l (Functional)			
Heterogeneity: Chi ² = 1.85, df =	= I (P = 0.17); I ² =46%			
Test for overall effect: $Z = 1.97$ ((P = 0.049)			
2 Intermediate term				
Dettori 1994	1/18	0/46		7.42 [0.32, 174.21]
Eiff 1994	0/37	0/40		0.0 [0.0, 0.0]
Konradsen 1991	4/37	5/36		0.78 [0.23, 2.67]
Subtotal (95% CI)	92	122	-	1.14 [0.39, 3.27]
Total events: 5 (Immobilisation),	5 (Functional)			
Heterogeneity: Chi ² = 1.72, df =	= I (P = 0.19); I ² =42%			
Test for overall effect: $Z = 0.24$ ((P = 0.81)			
3 Long term				
Eiff 1994	0/37	0/40		0.0 [0.0, 0.0]
Subtotal (95% CI)	37	40		0.0 [0.0, 0.0]
Total events: 0 (Immobilisation),	0 (Functional)			
Heterogeneity: not applicable				
Test for overall effect: $Z = 0.0$ (P	<i>P</i> < 0.00001)			
			0.1 0.2 0.5 2 5 10	

Favours immobilis. Favours functional
Analysis I.3. Comparison I Immobilisation vs. functional treatment, Outcome 3 Pain.

Review: Immobilisation and functional treatment for acute lateral ankle ligament injuries in adults

Comparison: I Immobilisation vs. functional treatment

Outcome: 3 Pain

Study or subgroup	Immobilisation n/N	Functional n/N	Risk Ratio M-H,Fixed,95% Cl	Risk Ratio M-H,Fixed,95% Cl
I Short term				
Cetti 1984	8/40	0/40		17.00 [1.01, 284.96]
Dettori 1994	4/16	14/40		0.71 [0.28, 1.84]
Eiff 1994	23/37	19/40		1.31 [0.87, 1.97]
Subtotal (95% CI)	93	120	•	1.42 [0.97, 2.09]
Total events: 35 (Immobilisation Heterogeneity: $Chi^2 = 5.17$, df Test for overall effect: $Z = 1.81$ 2 Intermediate term	$= 2 (P = 0.08); I^2 = 6 I \%$			
Cetti 1984	3/40	0/40		7.00 [0.37, 131.28]
Dettori 1994	4/ 8	14/46		0.73 [0.28, 1.92]
Eiff 1994	4/37	4/40	_	1.08 [0.29, 4.01]
Hedges 1980	9/14	9/17		1.21 [0.67, 2.20]
Lind 1984	3/29	0/27		6.53 [0.35, 120.90]
Subtotal (95% CI) Total events: 23 (Immobilisation Heterogeneity: $Chi^2 = 3.87$, df Test for overall effect: $Z = 1.00$ 3 Long term	$= 4 (P = 0.42); I^2 = 0.0\%$	170		1.28 [0.79, 2.06]
Eiff 1994	0/37	0/40		0.0 [0.0, 0.0]
Freeman 1965a	7/17	5/12		0.99 [0.41, 2.38]
Gronmark 1978	11/33	7/30	_ _	1.43 [0.64, 3.21]
Klein 1991	8/27	9/27	_ _	0.89 [0.40, 1.96]
Munk 1995	4/44	0/16		3.40 [0.19, 59.84]
Subtotal (95% CI) Total events: 30 (Immobilisation Heterogeneity: $Chi^2 = 1.37$, df Test for overall effect: $Z = 0.64$	$= 3 (P = 0.71); I^2 = 0.0\%$	125		1.17 [0.73, 1.87]
			0.1 0.2 0.5 1 2 5 10	
			Favours immobilis. Favours functional	

Analysis I.4. Comparison I Immobilisation vs. functional treatment, Outcome 4 Swelling.

Review: Immobilisation and functional treatment for acute lateral ankle ligament injuries in adults

Comparison: I Immobilisation vs. functional treatment

Outcome: 4 Swelling

Study or subgroup	Immobilisation n/N	Functional n/N	Risk Ratio M-H,Fixed,95% Cl	Risk Ratio M-H,Fixed,95% CI
I Short term				
Brakenbury 1983	3/67	2/36		0.81 [0.14, 4.60]
Cetti 1984	25/40	10/40		2.50 [1.39, 4.50]
Eiff 1994	16/37	13/40		1.33 [0.75, 2.38]
Subtotal (95% CI)	144	116	•	1.74 [1.17, 2.59]
Total events: 44 (Immobilisation), 25 Heterogeneity: Chi ² = 3.03, df = 2 Test for overall effect: Z = 2.74 (P = 2 Intermediate term	$(P = 0.22); I^2 = 34\%$			
Eiff 1994	1/37	1/40	←	1.08 [0.07, 16.67]
Subtotal (95% CI) Total events: (Immobilisation), (F Heterogeneity: not applicable Test for overall effect: Z = 0.06 (P = 3 Long term	,	40		1.08 [0.07, 16.67]
Eiff 1994	0/37	0/40		0.0 [0.0, 0.0]
Freeman 1965a	6/17	5/12		0.85 [0.33, 2.14]
Klein 1991	0/27	1/27	· • •	0.33 [0.01, 7.84]
Moller-Larsen 1988	10/55	9/65		1.31 [0.57, 3.00]
Subtotal (95% CI)	136	144	-	1.04 [0.57, 1.91]
Total events: 16 (Immobilisation), 15 Heterogeneity: $Chi^2 = 0.99$, df = 2 Test for overall effect: Z = 0.14 (P =	$(P = 0.6 I); I^2 = 0.0\%$			

Favours immobilis. Favours functional

Analysis 1.5. Comparison I Immobilisation vs. functional treatment, Outcome 5 Subjective instability (giving way).

Review: Immobilisation and functional treatment for acute lateral ankle ligament injuries in adults

Comparison: I Immobilisation vs. functional treatment

Outcome: 5 Subjective instability (giving way)

Study or subgroup	Immobilisation	Functional	Risk Ratio	Weight	Risk Ratio
	n/N	n/N	M-H,Fixed,95% CI		M-H,Fixed,95% Cl
I Short term					
Eiff 1994	7/37	5/40		100.0 %	.5 [0.53, 4.36]
Subtotal (95% CI)	37	40		100.0 %	1.51 [0.53, 4.36]
Total events: 7 (Immobilisatio	n), 5 (Functional)				
Heterogeneity: not applicable	2				
Test for overall effect: $Z = 0.7$	77 (P = 0.44)				
2 Intermediate term					
Cetti 1984	12/40	6/40		38.4 %	2.00 [0.83, 4.81]
Dettori 1994	3/18	12/46		43.2 %	0.64 [0.20, 2.00]
Eiff 1994	3/37	3/40	_	18.4 %	1.08 [0.23, 5.03]
Subtotal (95% CI)	95	126	-	100.0 %	1.24 [0.67, 2.30]
Heterogeneity: $Chi^2 = 2.47$, or Test for overall effect: $Z = 0.6$	()	6			
3 Long term	57 (1 – 0.77)				
Brostrom 1966	17/81	18/102		33.7 %	1.19 [0.66, 2.16]
Eiff 1994	1/37	1/40	· · · · · · · · · · · · · · · · · · ·	2.0 %	1.08 [0.07, 16.67]
Freeman 1965a	7/17	5/12	_	12.4 %	0.99 [0.41, 2.38]
Klein 1991	4/27	3/27		6.3 %	1.33 [0.33, 5.40]
Korkala 1987	15/47	19/36		45.5 %	0.60 [0.36, 1.02]
Subtotal (95% CI)	209	217	•	100.0 %	0.91 [0.64, 1.28]
Total events: 44 (Immobilisati	ion), 46 (Functional)				
Heterogeneity: Chi ² = 3.47, o	df = 4 (P = 0.48); $I^2 = 0.05$	%			
Test for overall effect: $Z = 0.5$	57 (P = 0.57)				
			0.1 0.2 0.5 1 2 5 10		
			Favours Immobilis. Favours functional		

Analysis I.6. Comparison I Immobilisation vs. functional treatment, Outcome 6 Objective instability (talar tilt and/or ADS).

Review: Immobilisation and functional treatment for acute lateral ankle ligament injuries in adults

Comparison: I Immobilisation vs. functional treatment

Outcome: 6 Objective instability (talar tilt and/or ADS)

Risk Rat	Weight	Risk Ratio	Functional	Immobilisation	Study or subgroup
M-H,Fixed,95%		M-H,Fixed,95% Cl	n/N	n/N	
					2 Intermediate term
0.97 [0.14, 6.54	56.6 %		2/36	2/37	Konradsen 1991
0.31 [0.01, 7.33	43.4 %	• •	1/27	0/29	Lind 1984
0.69 [0.14, 3.34	100.0 %		63	66	Subtotal (95% CI)
				n), 3 (Functional)	Total events: 2 (Immobilisation
			6	$df = (P = 0.54); ^2 = 0.09$	Heterogeneity: Chi ² = 0.37, c
				47 (P = 0.64)	Test for overall effect: Z = 0.4
					3 Long term
0.88 [0.54, 1.44	72.0 %		25/80	21/76	Brostrom 1966
0.96 [0.41, 2.29	23.7 %		7/30	9/40	Korkala 1987
1.09 [0.12, 9.74	4.3 %		1/16	3/44	Munk 1995
0.91 [0.60, 1.39	100.0 %	•	126	160	Subtotal (95% CI)
				on), 33 (Functional)	Total events: 33 (Immobilisatio
			6	df = 2 (P = 0.97); $I^2 = 0.09$	Heterogeneity: Chi ² = 0.06, c
				43 (P = 0.67)	Test for overall effect: $Z = 0.4$

0.1 0.2 0.5 2 5 10 Favours Immobilis. Favours functional

Analysis 1.7. Comparison I Immobilisation vs. functional treatment, Outcome 7 Recurrent sprain.

Review: Immobilisation and functional treatment for acute lateral ankle ligament injuries in adults

Comparison: I Immobilisation vs. functional treatment

Outcome: 7 Recurrent sprain

,	Immobilisation	Functional	Risk Ratio	Weight	Risk Ratio
	n/N	n/N	M-H,Fixed,95% Cl		M-H,Fixed,95% C
I Short term					
Eiff 1994	1/37	3/40		100.0 %	0.36 [0.04, 3.31
Subtotal (95% CI)	37	40		100.0 %	0.36 [0.04, 3.31
Total events: I (Immobilisati					
Heterogeneity: not applicab					
Test for overall effect: $Z = 0$ 2 Intermediate term	.90 (P = 0.37)				
Cetti 1984	12/40	6/40		25.1 %	2.00 [0.83, 4.81
Dettori 1994	4/18	12/46	_	28.2 %	0.85 [0.32, 2.30
Eiff 1994	3/37	3/40	_	12.0 %	1.08 [0.23, 5.03
Konradsen 1991	3/37	5/36		21.2 %	0.58 [0.15, 2.27
Lind 1984	3/29	1/27		4.3 %	2.79 [0.31, 25.25
Sommer 1993	0/33	3/73	• • • · · · · · · · · · · · · · · · · ·	9.2 %	0.31 [0.02, 5.85
Sommer 1775	0/55				
Subtotal (95% CI) Total events: 25 (Immobilisa	194	262	-	100.0 %	1.14 [0.69, 1.90
Subtotal (95% CI) Total events: 25 (Immobilisa Heterogeneity: $Chi^2 = 4.24$, Test for overall effect: $Z = 0$	194 tion), 30 (Functional) df = 5 (P = 0.52); I ² =0.0		-	100.0 %	1.14 [0.69, 1.90
Subtotal (95% CI) Total events: 25 (Immobilisa Heterogeneity: $Chi^2 = 4.24$, Test for overall effect: $Z = 0$ 3 Long term	194 tion), 30 (Functional) df = 5 (P = 0.52); l ² =0.0 .52 (P = 0.60)	%	-		1.14 [0.69, 1.90
Subtotal (95% CI) Total events: 25 (Immobilisa Heterogeneity: Chi ² = 4.24, Test for overall effect: Z = 0 3 Long term Brostrom 1966	194 tion), 30 (Functional) df = 5 (P = 0.52); I ² =0.0 .52 (P = 0.60) I7/81	%	•	45.0 %	1.19 [0.66, 2.16
Subtotal (95% CI) Total events: 25 (Immobilisa Heterogeneity: $Chi^2 = 4.24$, Test for overall effect: $Z = 0$ 3 Long term	194 tion), 30 (Functional) df = 5 (P = 0.52); l ² =0.0 .52 (P = 0.60)	%	• •		
Subtotal (95% CI) Total events: 25 (Immobilisa Heterogeneity: Chi ² = 4.24, Test for overall effect: Z = 0 3 Long term Brostrom 1966	194 tion), 30 (Functional) df = 5 (P = 0.52); I ² =0.0 .52 (P = 0.60) I7/81	%	• •	45.0 %	1.19 [0.66, 2.16
Subtotal (95% CI) Total events: 25 (Immobilisa Heterogeneity: Chi ² = 4.24, Test for overall effect: Z = 0 3 Long term Brostrom 1966 Eiff 1994	194 tion), 30 (Functional) df = 5 (P = 0.52); I ² =0.0 .52 (P = 0.60) I 7/8 I 3/37	% 18/102 3/40		45.0 % 8.1 %	1.19 [0.66, 2.16 1.08 [0.23, 5.03
Subtotal (95% CI) Total events: 25 (Immobilisa Heterogeneity: Chi ² = 4.24, Test for overall effect: Z = 0 3 Long term Brostrom 1966 Eiff 1994 Klein 1991	194 tion), 30 (Functional) df = 5 (P = 0.52); l ² =0.0 .52 (P = 0.60) 17/81 3/37 10/27	% 18/102 3/40 8/27		45.0 % 8.1 % 22.6 %	1.19 [0.66, 2.16 1.08 [0.23, 5.03 1.25 [0.58, 2.68
Subtotal (95% CI) Total events: 25 (Immobilisa Heterogeneity: Chi ² = 4.24, Test for overall effect: Z = 0 3 Long term Brostrom 1966 Eiff 1994 Klein 1991 Korkala 1987	194 tion), 30 (Functional) df = 5 (P = 0.52); I ² =0.0 .52 (P = 0.60) I7/8 I 3/37 I0/27 I0/47	% 18/102 3/40 8/27 6/36		45.0 % 8.1 % 22.6 % 19.2 %	1.19 [0.66, 2.16 1.08 [0.23, 5.03 1.25 [0.58, 2.68 1.28 [0.51, 3.19 1.09 [0.12, 9.74
Subtotal (95% CI) Total events: 25 (Immobilisa Heterogeneity: Chi ² = 4.24, Test for overall effect: Z = 0 3 Long term Brostrom 1966 Eiff 1994 Klein 1991 Korkala 1987 Munk 1995	194 tion), 30 (Functional) df = 5 (P = 0.52); I ² =0.0 .52 (P = 0.60) I7/81 3/37 I0/27 I0/47 3/44	% 18/102 3/40 8/27 6/36 1/16		45.0 % 8.1 % 22.6 % 19.2 % 4.1 %	1.19 [0.66, 2.16 1.08 [0.23, 5.03 1.25 [0.58, 2.68 1.28 [0.51, 3.19 1.09 [0.12, 9.74 5.73 [0.25, 129.23
Subtotal (95% CI) Total events: 25 (Immobilisa Heterogeneity: Chi ² = 4.24, Test for overall effect: Z = 0 3 Long term Brostrom 1966 Eiff 1994 Klein 1991 Korkala 1987 Munk 1995 Regis 1995 Subtotal (95% CI) Total events: 44 (Immobilisa	194 tion), 30 (Functional) df = 5 (P = 0.52); I ² =0.0 .52 (P = 0.60) 17/81 3/37 10/27 10/47 3/44 1/10 246 tion), 36 (Functional)	% 18/102 3/40 8/27 6/36 1/16 0/20 241		45.0 % 8.1 % 22.6 % 19.2 % 4.1 % 1.0 %	1.19 [0.66, 2.16 1.08 [0.23, 5.03 1.25 [0.58, 2.68 1.28 [0.51, 3.19
Subtotal (95% CI) Total events: 25 (Immobilisa Heterogeneity: Chi ² = 4.24, Test for overall effect: Z = 0 3 Long term Brostrom 1966 Eiff 1994 Klein 1991 Korkala 1987 Munk 1995 Regis 1995 Subtotal (95% CI)	194 tion), 30 (Functional) df = 5 (P = 0.52); I ² =0.0 .52 (P = 0.60) I7/8I 3/37 I0/27 I0/47 3/44 I/10 246 tion), 36 (Functional) df = 5 (P = 0.96); I ² =0.0	% 18/102 3/40 8/27 6/36 1/16 0/20 241		45.0 % 8.1 % 22.6 % 19.2 % 4.1 % 1.0 %	1.19 [0.66, 2.16 1.08 [0.23, 5.03 1.25 [0.58, 2.68 1.28 [0.51, 3.19 1.09 [0.12, 9.74 5.73 [0.25, 129.23

Favours Immobilis. Favours functional

Analysis I.8. Comparison I Immobilisation vs. functional treatment, Outcome 8 Decreased ROM.

Review: Immobilisation and functional treatment for acute lateral ankle ligament injuries in adults

Comparison: I Immobilisation vs. functional treatment

Outcome: 8 Decreased ROM

Study or subgroup	Immobilisation n/N	Functional n/N	Risk Ratio M-H,Fixed,95% Cl	Weight	Risk Ratio M-H,Fixed,95% Cl
I Short term					
Brakenbury 1983	16/77	14/47		97.2 %	0.70 [0.38, 1.30]
Cetti 1984	10/40	0/40		2.8 %	21.00 [1.27, 346.66]
Subtotal (95% CI)	117	87	-	100.0 %	1.27 [0.73, 2.19]
Total events: 26 (Immobilisa	tion), 14 (Functional)				
Heterogeneity: Chi ² = 7.41	, df = 1 (P = 0.01); l ² =87	1%			
Test for overall effect: $Z = 0$	0.84 (P = 0.40)				
2 Intermediate term					
Cetti 1984	3/40	0/40		100.0 %	7.00 [0.37, 3 .28]
Subtotal (95% CI)	40	40		100.0 %	7.00 [0.37, 131.28]
Total events: 3 (Immobilisat	ion), 0 (Functional)				
Heterogeneity: not applicab	le				
Test for overall effect: $Z = 1$	I.30 (P = 0.19)				
			0.1 0.2 0.5 2 5 10		
			Favours Immobilis. Favours functiona	al	

Analysis 1.9. Comparison I Immobilisation vs. functional treatment, Outcome 9 Patient satisfaction.

Review: Immobilisation and functional treatment for acute lateral ankle ligament injuries in adults

Comparison: I Immobilisation vs. functional treatment

Outcome: 9 Patient satisfaction

Study or subgroup	Immobilisation n/N	Functional n/N	Risk Ratio M-H,Fixed,95% Cl	Weight	Risk Ratio M-H,Fixed,95% Cl
I Intermediate term					
Cetti 1984	7/40	2/40		10.5 %	3.50 [0.77, 15.83]
Roycroft 1983	3/21	0/22		2.6 %	7.32 [0.40, 33.66]
Subtotal (95% CI)	61	62		13.1 %	4.25 [1.12, 16.09]
Total events: 10 (Immobilisati	on), 2 (Functional)				
Heterogeneity: Chi ² = 0.20, o	$df = 1 (P = 0.66); I^2 = 0.0$	%			
Test for overall effect: $Z = 2.1$	I3 (P = 0.033)				
2 Long term					
Klein 1991	6/27	3/27		15.7 %	2.00 [0.56, 7.19]
Moller-Larsen 1988	11/55	10/65		48.1 %	1.30 [0.60, 2.83]
Munk 1995	12/44	3/16		23.1 %	1.45 [0.47, 4.50]
Subtotal (95% CI)	126	108	-	86.9 %	1.47 [0.83, 2.60]
Total events: 29 (Immobilisati	on), 16 (Functional)				
Heterogeneity: Chi ² = 0.32, o	df = 2 (P = 0.85); $I^2 = 0.0$	%			
Test for overall effect: $Z = 1.3$	32 (P = 0.19)				
Total (95% CI)	187	170	-	100.0 %	1.83 [1.09, 3.07]
Total events: 39 (Immobilisati	on), 18 (Functional)				
Heterogeneity: $Chi^2 = 2.5I$, o	df = 4 (P = 0.64); $I^2 = 0.0$	%			
Test for overall effect: $Z = 2.2$	29 (P = 0.022)				
			0.1 0.2 0.5 1 2 5 10		

Favours Immobilis. Favours functional

Analysis 1.10. Comparison I Immobilisation vs. functional treatment, Outcome 10 Return to sports (days).

Review: Immobilisation and functional treatment for acute lateral ankle ligament injuries in adults

Comparison: I Immobilisation vs. functional treatment

Outcome: 10 Return to sports (days)

Study or subgroup	Immobilisation N	Mean(SD)	Functional N	Mean(SD)		an Difference ed,95% Cl	Weight	Mean Difference IV,Fixed,95% Cl
Return to sports (days)								
Freeman 1965a	17	154 (56)	12	84 (31.5)			1.1 %	70.00 [37.96, 102.04]
Milford 1990	30	22.9 ()	30	18.6 (8)			48.1 %	4.30 [-0.57, 9.17]
Sommer 1993	33	28 (10.5)	73	24 (13.5)			50.8 %	4.00 [-0.74, 8.74]
Subtotal (95% CI)	80		115			-	100.0 %	4.88 [1.50, 8.25]
Heterogeneity: $Chi^2 = 16$	6.06, df = 2 (P = 0.00)	0033); I ² =88%						
Test for overall effect: Z =	= 2.83 (P = 0.0046)							
Test for subgroup differer	nces: Not applicable							
					-10 -5	0 5 10)	
				Favo	urs Immobilis.	Favours funct	ional	

Analysis 1.11. Comparison I Immobilisation vs. functional treatment, Outcome 11 Return to work (days).

Review: Immobilisation and functional treatment for acute lateral ankle ligament injuries in adults

Comparison: I Immobilisation vs. functional treatment

Outcome: II Return to work (days)

Study or subgroup	Immobilisation		Functional		Mea	n Difference	Weight	Mean Difference
	Ν	Mean(SD)	Ν	Mean(SD)	IV,Fixe	ed,95% Cl		IV,Fixed,95% CI
I Return to work (days)								
Brakenbury 1983	75	16.3 (6.9)	43	11.6 (7.2)			52.5 %	4.70 [2.04, 7.36]
Brostrom 1966	81	34 (30)	102	8 ()		-	7.9 %	16.00 [9.13, 22.87]
Caro 1964	37	25.3 (17)	42	20 (30.5)			3.2 %	5.30 [-5.43, 6.03]
Cetti 1984	40	37.5 (15.3)	40	2.2 ()			10.9 %	25.30 [19.46, 31.14]
Dettori 1994	18	31.5 (33)	46	33.5 (16)	۰		1.5 %	-2.00 [-17.93, 13.93]
Roycroft 1983	37	18.6 (10)	43	11.9 (7.5)		∎ →	24.1 %	6.70 [2.77, 10.63]
Subtotal (95% CI)	288		316			•	100.0 %	8.23 [6.31, 10.16]
Heterogeneity: $Chi^2 = 46$	6.96, df = 5 (P<0.00	00 l); l ² =89%						
Test for overall effect: Z =	= 8.38 (P < 0.0000))						
Test for subgroup differer	nces: Not applicable							
					-10 -5 (0 5 10		

Favours Immobilis. Favours functional

Analysis 1.12. Comparison I Immobilisation vs. functional treatment, Outcome 12 Improvement in pain score.

Review: Immobilisation and functional treatment for acute lateral ankle ligament injuries in adults

Comparison: I Immobilisation vs. functional treatment Outcome: 12 Improvement in pain score Study or subgroup Immobilisation Functional Mean Difference Mean Difference Mean(SD) IV,Fixed,95% CI IV,Fixed,95% CI Ν Mean(SD) Ν I Short term Hedges 1980 49 -2.08 (1.26) 44 -1.84 (1.22) -0.24 [-0.74, 0.26] 0 10 -10 -5 5 Favours Immobilis. Favours functional

Analysis 1.13. Comparison I Immobilisation vs. functional treatment, Outcome 13 Improvement in swelling.

Review: Immobilisation and functional treatment for acute lateral ankle ligament injuries in adults

Comparison: I Immobilisation vs. functional treatment

Outcome: 13 Improvement in swelling

Study or subgroup	Immobilisation N	Mean(SD)	Functional N	Mean Mean(SD) IV,Fixed			Differenc 95% Cl	e	Mean Difference IV,Fixed,95% Cl	
l Short term Hedges 1980	49	-0.51 (0.58)	44	-0.34 (0.64)		_	+			-0.17 [-0.42, 0.08]
					-10 Favours in	-5 nmobilis.	0	5 Favours	10 functional	

Analysis 1.14. Comparison I Immobilisation vs. functional treatment, Outcome 14 Improvement in objective instability (difference in TT).

Review: Immobilisation and functional treatment for acute lateral ankle ligament injuries in adults

Comparison: I Immobilisation vs. functional treatment

Outcome: 14 Improvement in objective instability (difference in TT)

Study or subgroup	Immobilisation		Functional		Mea	n Difference	Mean Difference
	Ν	Mean(SD)	Ν	Mean(SD)	IV,Rand	om,95% Cl	IV,Random,95% CI
2 Intermediate term Sommer 1993	33	-7.6 (3.1)	73	-10.2 (3.75)			2.60 [1.24, 3.96]
3 Long term Freeman 1965a	17	-14.72 (10.77)	12	-9.45 (4.8)	<u>، ا</u>	_	-5.27 [-11.07, 0.53]
					-10 -5 Favours imobilis.	0 5 10 Favours functional	

Analysis 1.15. Comparison I Immobilisation vs. functional treatment, Outcome 15 Improvement in ROM.

Review: Immobilisation and functional treatment for acute lateral ankle ligament injuries in adults

Comparison: I Immobilisation vs. functional treatment

Outcome: 15 Improvement in ROM

Study or subgroup	Immobilisation		Functional			Μ	lean	Differenc	e	Mean Difference
	Ν	Mean(SD)	Ν	Mean(SD)		IV,Fi	ixed,	95% CI		IV,Fixed,95% CI
I Short term										
Brakenbury 1983	75	10.4 (2.2)	43	9.5 (3.9)			+	_		0.90 [-0.37, 2.17]
3 Long term										
Munk 1995	18	51 (6.6)	44	52 (6.4)			+	_		-1.00 [-4.59, 2.59]
					-10	-5	0	5	10	
				Fa	avours In	nmobilis.		Favours	functional	

Analysis 2.1. Comparison 2 Immobilisation vs. physiotherapy, Outcome I Return to work (days).

Review: Immobilisation and functional treatment for acute lateral ankle ligament injuries in adults

Comparison: 2 Immobilisation vs. physiotherapy

Outcome: I Return to work (days)

Study or subgroup	Immobilisation N	Mean(SD)	Physiotherapy N	Mean(SD)				Differend 95% Cl	ce	Mean Difference IV,Fixed,95% Cl
	IN	riean(SD)	IN	(SD)		IV,F	ixea,	,73 ⁄o CI		IV,FIXED,73% CI
I Short term										
Brooks 1981	26	14 (0)	21	6 (0)						0.0 [0.0, 0.0]
						1				
					-10	-5	0	5	10	
					Favours Ir	nmobilis.		Favours	physiother:	

Analysis 3.1. Comparison 3 Immobilisation vs. other types of immobilisation, Outcome I Pain.

Review: Immobilisation and functional treatment for acute lateral ankle ligament injuries in adults

Comparison: 3 Immobilisation vs. other types of immobilisation

Outcome: I Pain

Study or subgroup	Cast	Soft-cast	Risk Ratio	Risk Ratio
	n/N	n/N	M-H,Fixed,95% Cl	M-H,Fixed,95% Cl
I Short term				
Avci 1998	7/26	4/31		2.09 [0.69, 6.35]
			0.1 0.2 0.5 1 2 5 10	
			Favours cast Favours soft-cast	

Analysis 3.2. Comparison 3 Immobilisation vs. other types of immobilisation, Outcome 2 Swelling.

Review: Immobilisation and functional treatment for acute lateral ankle ligament injuries in adults

Study or subgroup	Cast	Soft-cast	Risk Ratio	Risk Ratio
	n/N	n/N	M-H,Fixed,95% Cl	M-H,Fixed,95% C
I Short term				
Avci 1998	12/26	9/31		1.59 [0.80, 3.17
			0.1 0.2 0.5 1 2 5 10	
			Favours cast Favours Soft-cast	

Analysis 3.3. Comparison 3 Immobilisation vs. other types of immobilisation, Outcome 3 Objective instability (talar tilt and/or ADS).

Review: Immobilisation and functional treatment for acute lateral ankle ligament injuries in adults

Comparison: 3 Immobilisation vs. other types of immobilisation

Outcome: 3 Objective instability (talar tilt and/or ADS)

Study or subgroup	Cast n/N	Soft-cast n/N	Risk Ratio M-H,Fixed,95% Cl	Risk Ratio M-H,Fixed,95% Cl
l Short term Avci 1998	2/26	4/31		0.60 [0.12, 3.00]
			0.1 0.2 0.5 1 2 5 10 Favours cast Favours Soft-cast	

Analysis 3.4. Comparison 3 Immobilisation vs. other types of immobilisation, Outcome 4 Return to work (days).

Review: Immobilisation and functional treatment for acute lateral ankle ligament injuries in adults

Comparison: 3 Immobilisation vs. other types of immobilisation

Outcome: 4 Return to work (days)

Study or subgroup	Cast N	Mean(SD)	Soft-cast N	Mean(SD)		an Difference ed,95% Cl	Mean Difference IV,Fixed,95% C
I Short term							
Avci 1998	17	6.3 (4.6)	19	2.5 (3.3)			3.80 [1.16, 6.44]
					-10 -5 Favours cast	0 5 10 Favours Soft-cast	
					Favour's Casi	Favour's Solt-Cast	

Analysis 4.1. Comparison 4 Immobilisation vs. functional treatment (high quality trials), Outcome I Numbers not returning to sports.

Review: Immobilisation and functional treatment for acute lateral ankle ligament injuries in adults

Comparison: 4 Immobilisation vs. functional treatment (high quality trials)

Outcome: I Numbers not returning to sports

Study or subgroup	Immobilisation	Functional	Risk Ratio	Risk Ratio
	n/N	n/N	M-H,Fixed,95% Cl	M-H,Fixed,95% CI
Short term				
Eiff 1994	7/37	7/40		1.08 [0.42, 2.79]
Regis 1995	1/10	8/20	← ■ ────	0.25 [0.04, 1.73]
Subtotal (95% CI)	47	60	-	0.71 [0.31, 1.63]
otal events: 8 (Immobilisation), 15	(Functional)			
Heterogeneity: $Chi^2 = 1.87$, df = 1	(P = 0.17); I ² =46%			
est for overall effect: $Z = 0.80$ (P	= 0.42)			
Intermediate term				
Eiff 1994	1/37	0/40		3.24 [0.14, 77.06]
Subtotal (95% CI)	37	40		3.24 [0.14, 77.06]
otal events: I (Immobilisation), 0	(Functional)			
leterogeneity: not applicable				
est for overall effect: $Z = 0.73$ (P	= 0.47)			
Long term				
Eiff 1994	0/37	0/40		0.0 [0.0, 0.0]
Klein 1991	16/26	5/24		2.95 [1.28, 6.82]
Korkala 1987	7/47	8/36		0.67 [0.27, 1.68]
Regis 1995	3/10	1/20		6.00 [0.71, 50.59]
Subtotal (95% CI)	120	120	•	1.70 [0.98, 2.95]
otal events: 26 (Immobilisation), I	4 (Functional)			
Heterogeneity: $Chi^2 = 6.98$, $df = 2$	2 (P = 0.03); I ² =71%			
est for overall effect: $Z = 1.90$ (P	= 0.057)			

Favours Immobilis.

Favours functional

Analysis 4.2. Comparison 4 Immobilisation vs. functional treatment (high quality trials), Outcome 2 Numbers not returning to work.

Review: Immobilisation and functional treatment for acute lateral ankle ligament injuries in adults

Comparison: 4 Immobilisation vs. functional treatment (high quality trials)

Outcome: 2 Numbers not returning to work

Study or subgroup	Immobilisation	Functional	F	Risk Ratio	Risk Ratio
	n/N	n/N	M-H,Fix	ed,95% Cl	M-H,Fixed,95% Cl
I Short term					
Eiff 1994	1/37	1/40	•	• •	1.08 [0.07, 16.67]
2 Intermediate term					
Eiff 1994	0/37	0/40			0.0 [0.0, 0.0]
3 Long term					
Eiff 1994	0/37	0/40			0.0 [0.0, 0.0]
			0.1 0.2 0.5	2 5 10	
			Favours Immobilis.	Favours functional	

Analysis 4.3. Comparison 4 Immobilisation vs. functional treatment (high quality trials), Outcome 3 Pain.

Review: Immobilisation and fu	unctional treatment for acute la	ateral ankle ligament injur	ies in adults	
Comparison: 4 Immobilisation	n vs. functional treatment (high	quality trials)		
Outcome: 3 Pain				
Study or subgroup	Immobilisation	Functional	Risk Ratio	Risk Ratio
	n/N	n/N	M-H,Fixed,95% Cl	M-H,Fixed,95% Cl
I Short term				
Eiff 1994	23/37	19/40	-	1.31 [0.87, 1.97]
Subtotal (95% CI)	37	40	-	1.31 [0.87, 1.97]
Total events: 23 (Immobilisation)), 19 (Functional)			
Heterogeneity: not applicable				
Test for overall effect: $Z = 1.28$	(P = 0.20)			
2 Intermediate term				
Eiff 1994	4/37	4/40		1.08 [0.29, 4.01]
Hedges 1980	9/14	9/17		1.21 [0.67, 2.20]
			0.1 0.2 0.5 1 2 5 10	
			Favours Immobilis. Favours functional	(-
				(Continued)

Study or subgroup	Immobilisation n/N	Functional n/N	Risk Ratio M-H,Fixed,95% Cl	(Continued Risk Ratio M-H,Fixed,95% Cl
Subtotal (95% CI)	51	57	*	1.17 [0.66, 2.09]
Total events: 13 (Immobilisation)	, I 3 (Functional)			
Heterogeneity: Chi ² = 0.03, df =	= I (P = 0.87); I ² =0.0%			
Test for overall effect: $Z = 0.54$	(P = 0.59)			
3 Long term				
Eiff 1994	0/37	0/40		0.0 [0.0, 0.0]
Gronmark 1978	/33	7/30		1.43 [0.64, 3.21]
Klein 1991	8/27	9/27		0.89 [0.40, 1.96]
Munk 1995	4/44	0/16		3.40 [0.19, 59.84]
Subtotal (95% CI)	141	113	-	1.23 [0.71, 2.14]
Total events: 23 (Immobilisation)	, I 6 (Functional)			
Heterogeneity: Chi ² = 1.26, df =	= 2 (P = 0.53); I ² =0.0%			
Test for overall effect: $Z = 0.73$	(P = 0.47)			
			0.1 0.2 0.5 1 2 5 10	
			Favours Immobilis. Favours functional	

Analysis 4.4. Comparison 4 Immobilisation vs. functional treatment (high quality trials), Outcome 4 Swelling.

Review: Immobilisation and functional treatment for acute lateral ankle ligament injuries in adults

Comparison: 4 Immobilisation vs. functional treatment (high quality trials)

Outcome: 4 Swelling

Study or subgroup	Immobilisation	Functional	Risk Ratio	Risk Ratio
	n/N	n/N	M-H,Fixed,95% Cl	M-H,Fixed,95% Cl
I Short term				
Eiff 1994	16/37	13/40		1.33 [0.75, 2.38]
Subtotal (95% CI)	37	40	-	1.33 [0.75, 2.38]
Total events: 16 (Immobilisation),	I 3 (Functional)			
Heterogeneity: not applicable				
Test for overall effect: $Z = 0.97$ (F	P = 0.33)			
2 Intermediate term				
Eiff 1994	1/37	1/40	← →	1.08 [0.07, 16.67]
Subtotal (95% CI)	37	40		1.08 [0.07, 16.67]
Total events: (Immobilisation),	(Functional)			
Heterogeneity: not applicable				
Test for overall effect: $Z = 0.06$ (F	P = 0.96)			
3 Long term				
Eiff 1994	0/37	0/40		0.0 [0.0, 0.0]
Klein 1991	0/27	1/27	← <mark>■</mark>	0.33 [0.01, 7.84]
Subtotal (95% CI)	64	67		0.33 [0.01, 7.84]
Total events: 0 (Immobilisation), I	(Functional)			
Heterogeneity: $Chi^2 = 0.0$, $df = 0$	$(P = 1.00); I^2 = 0.0\%$			
Test for overall effect: $Z = 0.68$ (F	P = 0.50)			
			0.1 0.2 0.5 1 2 5 10	
			Favours Immobilis. Favours functional	

Analysis 4.5. Comparison 4 Immobilisation vs. functional treatment (high quality trials), Outcome 5 Subjective instability (giving way).

Review: Immobilisation and functional treatment for acute lateral ankle ligament injuries in adults

Comparison: 4 Immobilisation vs. functional treatment (high quality trials)

Outcome: 5 Subjective instability (giving way)

Study or subgroup	Immobilisation	Functional	Risk Ratio	Weight	Risk Ratio
	n/N	n/N	M-H,Fixed,95% CI		M-H,Fixed,95% Cl
I Short term					
Eiff 1994	7/37	5/40		100.0 %	.5 [0.53, 4.36]
Subtotal (95% CI)	37	40		100.0 %	1.51 [0.53, 4.36]
Total events: 7 (Immobilisatio	n), 5 (Functional)				
Heterogeneity: not applicable	:				
Test for overall effect: $Z = 0.7$	77 (P = 0.44)				
2 Intermediate term					
Eiff 1994	3/37	3/40		100.0 %	1.08 [0.23, 5.03]
Subtotal (95% CI)	37	40		100.0 %	1.08 [0.23, 5.03]
Total events: 3 (Immobilisation	n), 3 (Functional)				
Heterogeneity: not applicable	:				
Test for overall effect: $Z = 0.1$	0 (P = 0.92)				
3 Long term					
Brostrom 1966	17/81	18/102	-	38.5 %	1.19 [0.66, 2.16]
Eiff 1994	1/37	1/40	· · · · · · · · · · · · · · · · · · ·	2.3 %	1.08 [0.07, 16.67]
Klein 1991	4/27	3/27		7.2 %	1.33 [0.33, 5.40]
Korkala 1987	15/47	19/36		52.0 %	0.60 [0.36, 1.02]
Subtotal (95% CI)	192	205	•	100.0 %	0.89 [0.62, 1.30]
Total events: 37 (Immobilisati	on), 41 (Functional)				
Heterogeneity: $Chi^2 = 3.39$, c	$f = 3 (P = 0.34); I^2 = I I^9$	6			
Test for overall effect: $Z = 0.5$	59 (P = 0.55)				
			0.1 0.2 0.5 2 5 10		
		1	Favours Immobilis. Favours functiona	1	

Analysis 4.6. Comparison 4 Immobilisation vs. functional treatment (high quality trials), Outcome 6 Objective instability (talar tilt and/or ADS).

Review: Immobilisation and functional treatment for acute lateral ankle ligament injuries in adults

Comparison: 4 Immobilisation vs. functional treatment (high quality trials)

Outcome: 6 Objective instability (talar tilt and/or ADS)

Study or subgroup	Immobilisation	Functional	Risk Ratio	Weight	Risk Ratio
	n/N	n/N	M-H,Fixed,95% CI		M-H,Fixed,95% Cl
3 Long term					
Brostrom 1966	21/76	25/80		72.0 %	0.88 [0.54, 1.44]
Korkala 1987	9/40	7/30	_ -	23.7 %	0.96 [0.41, 2.29]
Munk 1995	3/44	1/16		4.3 %	1.09 [0.12, 9.74]
Subtotal (95% CI)	160	126	•	100.0 %	0.91 [0.60, 1.39]
Total events: 33 (Immobilisa	tion), 33 (Functional)				
Heterogeneity: Chi ² = 0.06,	df = 2 (P = 0.97); I ² =0.0%	6			
Test for overall effect: $Z = 0$.43 (P = 0.67)				
			0.1 0.2 0.5 2 5 10		

Favours Immobilis. Favours functional

Analysis 4.7. Comparison 4 Immobilisation vs. functional treatment (high quality trials), Outcome 7 Recurrent sprain.

Review: Immobilisation and functional treatment for acute lateral ankle ligament injuries in adults

Comparison: 4 Immobilisation vs. functional treatment (high quality trials)

Outcome: 7 Recurrent sprain

Study or subgroup	Immobilisation	Functional	Risk Ratio	Weight	Risk Ratio
	n/N	n/N	M-H,Fixed,95% Cl		M-H,Fixed,95% CI
I Short term					
Eiff 1994	1/37	3/40		100.0 %	0.36 [0.04, 3.31]
Subtotal (95% CI)	37	40		100.0 %	0.36 [0.04, 3.31]
Total events: I (Immobilisatio	on), 3 (Functional)				
Heterogeneity: not applicable					
Test for overall effect: $Z = 0.9$	90 (P = 0.37)				
2 Intermediate term					
Eiff 1994	3/37	3/40		100.0 %	1.08 [0.23, 5.03]
Subtotal (95% CI)	37	40		100.0 %	1.08 [0.23, 5.03]
Total events: 3 (Immobilisatio	on), 3 (Functional)				
Heterogeneity: not applicable	2				
Test for overall effect: $Z = 0$.	10 (P = 0.92)				
3 Long term					
Brostrom 1966	17/81	18/102		45.0 %	1.19 [0.66, 2.16]
Eiff 1994	3/37	3/40		8.1 %	1.08 [0.23, 5.03]
Klein 1991	10/27	8/27		22.6 %	1.25 [0.58, 2.68]
Korkala 1987	10/47	6/36		19.2 %	1.28 [0.51, 3.19]
Munk 1995	3/44	1/16		4.1 %	1.09 [0.12, 9.74]
Regis 1995	1/10	0/20		1.0 %	5.73 [0.25, 129.23]
Subtotal (95% CI)	246	241	*	100.0 %	1.25 [0.84, 1.85]
Total events: 44 (Immobilisati	ion), 36 (Functional)				
Heterogeneity: $Chi^2 = 0.99$,	df = 5 (P = 0.96); $I^2 = 0.06$	%			
Test for overall effect: $Z = I$.	12 (P = 0.26)				
			<u> </u>		
			0.1 0.2 0.5 1 2 5 10		
			Favours Immobilis. Favours functional		

Analysis 4.8. Comparison 4 Immobilisation vs. functional treatment (high quality trials), Outcome 8 Patient satisfaction.

Review: Immobilisation and functional treatment for acute lateral ankle ligament injuries in adults

Comparison: 4 Immobilisation vs. functional treatment (high quality trials)

Outcome: 8 Patient satisfaction

Study or subgroup	Immobilisation	Functional	Risk Ratio	Weight	Risk Ratio
	n/N	n/N	M-H,Fixed,95% CI		M-H,Fixed,95% Cl
3 Long term					
Klein 1991	6/27	3/27		40.5 %	2.00 [0.56, 7.19]
Munk 1995	12/44	3/16		59.5 %	1.45 [0.47, 4.50]
Subtotal (95% CI)	71	43		100.0 %	1.68 [0.72, 3.90]
Total events: 18 (Immobilisat	tion), 6 (Functional)				
Heterogeneity: Chi ² = 0.13,	df = (P = 0.7); $ ^2 = 0.0\%$	6			
Test for overall effect: $Z = 1$.	.20 (P = 0.23)				
			0.1 0.2 0.5 1 2 5 10		
			Favours Immobilis. Favours function	al	

Analysis 4.9. Comparison 4 Immobilisation vs. functional treatment (high quality trials), Outcome 9 Return to work (days).

Review: Immobilisation and functional treatment for acute lateral ankle ligament injuries in adults

Comparison: 4 Immobilisation vs. functional treatment (high quality trials)

Outcome: 9 Return to work (days)

Study or subgroup	Immobilisation N	Mean(SD)	Functional N	Mean(SD)		n Difference :d,95% Cl	Weight	Mean Difference IV,Fixed,95% Cl
I Return to work (days)								
Brostrom 1966	81	34 (30)	102	18 (11)		-	70.9 %	16.00 [9.13, 22.87]
Caro 1964	37	25.3 (17)	42	20 (30.5)	—		29.1 %	5.30 [-5.43, 16.03]
Subtotal (95% CI)	118		144			_	100.0 %	12.89 [7.10, 18.67]
Heterogeneity: $Chi^2 = 2.7$	'I, df = I (P = 0.10); I ² =63%						
Test for overall effect: $Z =$	4.36 (P = 0.00001	3)						
Test for subgroup differen	ces: Not applicable							
					-5		2	
				-10		0 5 10		
				Favours	mmobilis.	Favours funct	ional	

Analysis 4.10. Comparison 4 Immobilisation vs. functional treatment (high quality trials), Outcome 10 Improvement in pain score.

Review: Immobilisation and functional treatment for acute lateral ankle ligament injuries in adults

Comparison: 4 Immobilisation vs. functional treatment (high quality trials)

Outcome: 10 Improvement in pain score

Study or subgroup	Immobilisation N	Mean(SD)	Functional N	Mean(SD)		n Difference d,95% Cl	Mean Difference IV,Fixed,95% Cl
Short term Hedges 1980	49	-2.08 (1.26)	44	-1.84 (1.22)	-	•	-0.24 [-0.74, 0.26]
					-10 -5 Favours Immobilis.	0 5 IO Favours functic	onal

Analysis 4.11. Comparison 4 Immobilisation vs. functional treatment (high quality trials), Outcome 11 Improvement in swelling.

Review: Immobilisation and functional treatment for acute lateral ankle ligament injuries in adults Comparison: 4 Immobilisation vs. functional treatment (high quality trials) Outcome: II Improvement in swelling Mean Difference Mean Difference Immobilisation Functional Study or subgroup IV.Fixed.95% CI IV,Fixed,95% CI Mean(SD) Mean(SD) Ν Ν I Short term Hedges 1980 49 -0.51 (0.58) 44 -0.34 (0.64) -0.17 [-0.42, 0.08] 0 -10 -5 5 10 Favours Immobilis. Favours functional

Analysis 4.12. Comparison 4 Immobilisation vs. functional treatment (high quality trials), Outcome 12 Improvement in ROM.

Review: Immobilisation and functional treatment for acute lateral ankle ligament injuries in adults

Comparison: 4 Immobilisation vs. functional treatment (high quality trials)

Outcome: 12 Improvement in ROM

Study or subgroup	Immobilisation N	Mean(SD)	Functional N	Mean(SD)				Differenc 95% Cl	e	Mean Difference IV,Fixed,95% CI
3 Long term Munk 1995	18	51 (6.6)	44	52 (6.4)			•	_		-1.00 [-4.59, 2.59]
					-10	-5	0	5	10	
					Favours Ir	nmobilis		Favours	functional	

ADDITIONAL TABLES

Table 1. Quality assessment tool

Item	Score	Notes
A. Was the assigned treatment adequately concealed prior to allocation?	 2 = method did not allow disclosure of assignment 1 = small but possible chance of disclosure of assignment or unclear 0 = quasi-randomised or open list/tables 	Cochrane code: clearly yes = A, not sure = B, clearly no = C
B. Were the outcomes of patients who with- drew described and included in the analysis (intention-to-treat)?		
C. Were the outcome assessors blinded to treatment status?	 2 = effective action taken to blind assessors 1 = small or moderate chance of unblinding of assessors 0 = not mentioned or not possible 	
D. Were the treatment and control group comparable at entry?	 2 = good comparability of groups, or confounding adjusted for in analysis 1 = confounding small; mentioned but not adjusted for 0 = large potential for confounding, or not discussed 	

Table 1. Quality assessment tool (Continued)

E. Were the subjects blind to assignment status after allocation?	 2 = effective action taken to blind subjects 1 = small or moderate chance of unblinding subjects 0 = not possible, or not mentioned (unless double-blind), or possible but not done 	
F. Were the treatment providers blind to assignment status after allocation?	2 = effective action taken to blind treatment providers 1 = small or moderate chance of unblinding of treatment providers 0 = not possible, or not mentioned (unless double-blind), or possible but not done	
G. Were care programmes, other than the trial options, identical?	 2 = care programmes clearly identical 1 = clear but trivial differences 0 = not mentioned or clear and important differences in care programmes 	
H. Were the inclusion and exclusion crite- ria clearly defined?	2 = clearly defined 1 = inadequately defined 0 = not defined	
J. Were the outcome measures used clearly defined?	2 = clearly defined 1 = inadequately defined 0 = not defined	
K. Was follow-up active and appropriate?	2 = optimal 1 = adequate 0 = not defined, not adequate	
L. Was the duration of surveillance clini- cally appropriate?	2 = optimal (short-intermediate-long)1 = adequate (short-intermediate) 0 = not defined, not adequate	

APPENDICES

Appendix I. Search strategy for MEDLINE (OVID Web)

Ankle Injuries/
 Ligaments, Articular/
 "Sprains and Strains"/
 or/1-3
 ankle\$.tw.
 ligament\$.tw.
 and/5-6
 (sprain\$ or strain\$ or injur\$ or rupture\$ or tear or torn).tw.
 and/7-8
 and/4,9
 Lateral Ligament, Ankle/
 or/10-11

WHAT'S NEW

Last assessed as up-to-date: 13 May 2002.

Date	Event	Description
5 November 2008	Amended	Converted to new review format.

HISTORY

Protocol first published: Issue 4, 1999

Review first published: Issue 3, 2002

CONTRIBUTIONS OF AUTHORS

This review was initiated by Gino Kerkhoffs (GK), Pim Assendelft (WA) and others (including Helen Handoll)(HH) and some preliminary work done. GK took the role of caretaker. All the results were checked and the analyses restructured to conform to the protocol. Initial trial location was performed by GK and Peter Struijs (PS) and subsequently by GK and Lesley Gillespie (LG). Study selection was done by at least two reviewers and always GK and PS. GK, BR, KK, PS and Niek van Dijk (CVD) participated in quality assessment and data extraction of the included trials. Compilation of the comparisons, structuring the review, data entry into RevMan, and composition of the drafts were shared by all authors. WA, BR, KK, CVD and PS advised on the analysis and content and provided critical feedback on the work at various stages. Gino Kerkhoffs is the guarantor of the review.

DECLARATIONS OF INTEREST

None known.

SOURCES OF SUPPORT

Internal sources

- Orthopaedic Research Center, Department of Orthopaedic Surgery, University of Amsterdam, Amsterdam, Netherlands.
- Division of Emergency Medicine, University of Alberta, Canada.

External sources

• No sources of support supplied

INDEX TERMS

Medical Subject Headings (MeSH)

*Immobilization; *Physical Therapy Modalities; Lateral Ligament, Ankle [*injuries]; Randomized Controlled Trials as Topic; Sprains and Strains [*therapy]

MeSH check words

Humans